

Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 16075

Permanent Instability of Preferences after COVID-19 Crisis: A Natural Experiment from Urban Burkina Faso

Delphine Boutin Laurene Petifour Haris Megzari

APRIL 2023



Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 16075

# Permanent Instability of Preferences after COVID-19 Crisis: A Natural Experiment from Urban Burkina Faso

#### **Delphine Boutin**

Bordeaux School of Economics, University of Bordeaux and IZA

#### Laurene Petifour

Heidelberg Institute of Global Health

### Haris Megzari

Bordeaux School of Economics and University of Bordeaux

APRIL 2023

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

# ABSTRACT

# Permanent Instability of Preferences after COVID-19 Crisis: A Natural Experiment from Urban Burkina Faso

The salience of the first Covid-19 crisis over a well-identified period makes it an unexpected and abrupt change in the environment. This study uses the onset of the Covid-19 crisis to empirically examine whether risk and time preferences change in response to this exogenous shock, and whether those variations are temporary or durable. We use an original panel dataset conducted in January 2020 (before any event), in June 2020 (just after the removal of strong economic measures) and in January 2022 among women working in the informal sector in Ouagadougou, Burkina Faso. We use individual fixed effects on the balanced panel to isolate the specific causal effect of the Covid-19 crisis on variation in attitudes toward risk and time. Two time horizons are analyzed: immediately at the end of the economic restrictions (short-term effect), then two years later (medium-term effect). We demonstrate strong preference instability: immediately after the shock of the Covid-19, risk aversion changed over the 6-month period in both the gain (12%) and loss (-47%) domains, while impatience increased by 20%. Eighteen months later, preferences have not returned to their pre-shock level, suggesting an abrupt and permanent effect of Covid-19 on individual preferences. We also show that risk aversion (in both domains) is non-sensitive to actual impacts, but appears to be driven by economic fears and concerns related to the Covid-19 crisis.

JEL Classification:	D8, D9, C93, I18, O55
Keywords:	COVID-19, risk attitudes, impatience, emotions,
	media exposure

## Corresponding author:

Delphine Boutin University of Bordeaux av. Leon Duguit 33600 Pessac France E-mail: delphine.boutin@u-bordeaux.fr

## 1 Introduction

In the spring of 2020, the shock of Covid-19 could be assimilated to a rare and sudden event, different from economic downturns or climatic shocks because it was completely unanticipable. In several Sub-Saharan countries, restrictive economic measures were strictly implemented as soon as the first cases appeared in the sub-region, so that the number of cases stayed really low during the first wave. In countries that maintained a low number of cases, the Covid-19 shock was mainly economic and emotional, although it represented an unprecedented and significant change in people's environment. Several articles have shown that the uncertainty and the considerable slowdown of the economy caused a decline in living standards and an increase in economic anxiety, especially at the beginning of the pandemic (Fetzer et al., 2021; Bargain and Aminjonov, 2021). In this paper, we show that the impact of the Covid-19 shock was severe, as it also affected permanently some deep parameters, notably risk and time preferences.

Contrary to the assumptions of standard microeconomic theory, empirical studies recently show that deep parameters (including main personality traits and individual preferences) can vary over time. A growing body of literature points out that these parameters are not immutable and can change due to economic, climatic, or conflict shocks (Biener and Landmann, 2023; Castillo and Hernandez, 2022; Fang et al., 2022; Chuang and Schechter, 2015; Cameron and Shah, 2015; Reynaud and Aubert, 2020; Callen, 2015; Voors et al., 2012; Hanaoka et al., 2018; Sakha, 2019). The results of this literature vary considerably from article to article, partly because of different samples, methodologies, and time frames. Several mechanisms come into play and can explain this instability. The most cited one relates to the adverse consequences of shocks, which modify people's points of reference and thus their risk attitudes (Kahneman and Tversky, 1979; Campbell and Cochrane, 1999). The Covid-19 crisis had a negative impact on the population's standard of living and generated greater risk aversion and impatience, particularly in the economic and financial fields. This change in risk behavior and time preference can translate to all domains, even those not affected by the shock in question. Gollier and Pratt (1996)'s "risk vulnerability" implies that in response to an increase in background risk (non-insurable and non-avoidable, as for the Covid-19 shock), individuals behave in a more risk-averse manner in other domains of avoidable risk. The second mechanism that may alter attitudes toward risk or time after Covid-19 is emotional (Lerner and Keltner, 2001; Loewenstein, 2000; Loewenstein and Lerner, 2003; Botzen et al., 2015; Eckel et al., 2009). Even if they have not experienced any loss, the fear (of contamination or job loss, for instance) and stress induced by economic and health uncertainty may alter risk and time preferences. Some studies have shown that the emotional channel may override more concrete explanations.

When the emotional response induces a shift in risk perception or discount rate, the behavioral change may persist over time (Lerner et al., 2015).

In this empirical study, we make use of an original survey to analyze whether and how risk and time preferences changed following the Covid-19 crisis in Ouagadougou, Burkina Faso. Providing insights to this issue is crucial as risk and time attitudes strongly influence a wide range of behaviors such savings, risk-sharing strategies, and migration decisions, determining various health and work outcomes (Barsky et al., 1997; Dohmen et al., 2011; Dawson and Henley, 2015; Hsieh et al., 2017; Van Der Pol et al., 2017; Kremer et al., 2019). Preference responsiveness after a shock or a change in the environment is potentially even higher in developing countries, where the capacity of public institutions and other compensatory mechanisms to absorb damaging impacts of shocks is often limited (Mosley and Verschoor, 2005; Kremer et al., 2019). Our study is based on a panel survey of women working in the informal sector The survey was first conducted in January 2020, before respondents were aware of the upcoming Covid crisis. Two time horizons are analyzed: immediately at the end of the economic restrictions (short-term effect), then two years later (medium-term effect). The short-term analysis (January-June 2020) is based on a balanced panel of 871 women interviewed twice. We reduced the medium-term analysis sample to 366 women interviewed three times (January 2020, June 2020, and January 2022). To elicit risk attitudes, we propose hypothetical gamble questions, similar to the Holt-Laury Paired Lottery Task (Holt and Laury, 2002). We capture time preferences using four sets of choice tasks, as Andersen et al. (2008) and Cassar et al. (2017). Our empirical identification strategy relies on a before-and-after comparison including individual fixed effects to isolate the variation in preferences in response to environmental changes between the baseline and the midline wave (short-term analysis), and the baseline and the endline wave (medium-term analysis). Therefore, we identify the change in attitudes toward risk and time both over a 6-month and a 2-year period. In the short-term analysis (6 months), we capture a causal effect of the Covid-19 crisis on preferences. Indeed, given the magnitude of the Covid-19 shock over the period, all characteristics that varied over the 6 months are directly or indirectly related to Covid-19. For the medium-term analysis (24 months), the results are very similar to the short term results, both in terms of significance and magnitude. Although we cannot completely rule out the role of other determinants of preferences shifts, we argue that the short term effects persisted over time and that we might capture long-lasting effects of the Covid-19 crisis.

We find a 12% increase in risk aversion when lotteries are presented in monetary gains in the short-term period. Eighteen months later, risk aversion has not returned to its pre-Covid level, as we see an 11% increase between the baseline level (January 2020) and the endline (January 2022),

once time-variant characteristics have been taken into account. Risk aversion instability is larger when lotteries are presented in terms of losses, with risk aversion decreasing by 47% in the short term. It remains at this level in the medium term, as we also see a 46% drop between the baseline and the endline. The reversal of the sign of the effect according to the framing is consistent with the reflection effect of Kahneman and Tversky (1979), which postulates that preferences depend on the reference, inducing risk aversion for gains and risk-seeking for losses. We also find that preference for present increased by 20% over the short-term period and by 15% over the mediumterm period. These results are robust to the different specifications. Then, we aim at revealing potential transmission channels through the identification of heterogeneous effects. We compare variations of preferences according to individuals' self-reports of their experience of the Covid-19 crisis (personal contamination, contamination of a relative), its impact on their daily life (recent loss of job, difficulties to meet the household basic needs) and their pandemic-related concerns. This suggestive evidence shows little effect of the direct consequences of Covid-19: preference instability is only amplified when the individual reports having difficulties with food or water access during the period, but is not affected by job loss nor by low-income levels. We find no different effect related to the respondent's health status (such as poor general health or having had Covid-19 symptoms during the study period). In contrast, preference instability is exacerbated when the respondent expresses concerns about the Covid-19 crisis (particularly regarding economic concerns and concerns related to more catastrophic scenarios such as economic collapse). We also capture the emotional mechanism through the exposition to different media. Indeed, media can reinforce the emotional response by increasing risk perception and anxiety (Shachat et al., 2021; Sadish et al., 2021; Fetzer et al., 2021). We do find that the more informed individuals exhibit wider change in their preferences over the study period. Unfortunately, we do not know the type of information assimilated, nor its veracity. However, the type of media used to keep up to date about the pandemic provides indications. We find that social networks as a primary source of information exacerbate preferences instability. In contrast, traditional media - television, radio, newspapers (which provide commentary and analysis on the situation and government announcements) and discussions with family and friends do not affect the impact on preference variation. This is in line with the results from Depoux et al. (2020): social networks (Facebook, Whatapps) are more likely to generate fear and anger than traditional media. Together, this set of findings suggests that the emotional channel is dominant in Covid-19-related variation in risk and time preferences.

We contribute to the literature in several ways. First, we complement the research stream that studies the stability of risk and time preferences after the experience of a shock. Our specific setting (a sudden and unanticipated shock, pre- and post-shock panel data) allows for a clean identification, which avoids the main empirical flaws of the literature. The latter has not reached a consensus (see next section), partly because of the difficulty of identifying the causal effect of shocks on preferences in a natural setting. Macroeconomic, natural disasters, and some conflicts (predominantly studied in the literature) can, to some extent, be anticipated or avoided (by migration to less risky locations or by insurance), which leads to selection bias-related issues. In this study, we observe the same individuals just before the Covid-19 pandemic, and six months after, once the various measures put in place at the beginning of the pandemic have been alleviated. Moreover, our short-term sample does not suffer from migration bias (no one migrated between the two waves) or selective attrition (we reached 95% of our respondents at midline and verified that there are no big differences between attritors and non-attritors both at midline). Our short-term analysis is then a causal one.

Second, we propose to study whether the instability observed in the short term persists over time, and we do find that preferences have not returned to their initial levels. To our knowledge, no paper has studied the variation in preferences at multiple points in time in the context of a developing country. These two-time analyses of the same individuals support the idea that exogenous shocks generate abrupt changes, which may be permanent. We acknowledge the fact that over the two years, many other changes, potentially independent of the consequences of Covid-19, may have maintained these new levels of preferences, and that this could affect our identification strategy.

Third, we contribute to the literature on Covid-19 related preference change (Castillo and Hernandez, 2022; Aragon et al., 2022; Angrisani et al., 2020; Drichoutis and Nayga, 2022; Gassmann et al., 2022; Bu et al., 2020; Li et al., 2020; Ikeda et al., 2020; Shachat et al., 2021). Most of these articles use small samples and very specific populations (mainly students or informed internet users) that may not consistently reflect behaviors in the field due to their singular characteristics. Furthermore, to our knowledge, no study has examined the impact of Covid-19 on preference stability in Africa. African countries were hit by Covid-19 with an unequal intensity, and in several countries few cases were recorded in 2020. For example, in Burkina Faso, the number of confirmed cases was 1000 and the number of deaths was 54 in July 2020.<sup>1</sup> The nature of the Covid-19 shock is different in these countries than in other contexts studied in the literature, such as China or Europe, where the economic crisis was mixed with the health crisis. In this respect, the context of low-impacted countries (regarding Covid-19 incidence) deserves specific investigation.

Fourth, this article contributes to the literature showing the importance of media and information in shaping a wide range of behaviors and beliefs (Dupas, 2011; La Ferrara et al., 2012; La Ferrara, 2016; Banerjee et al., 2019) and responses to Covid-19 perceptions and behaviors (Simonov et al.,

<sup>&</sup>lt;sup>1</sup>These numbers are most likely underestimated because only suspected cases and travelers were tested. The number of available tests was limited, and many Burkinabe people were reluctant to get tested because of the psychosis caused by the disease and the fear of the disease (Kobiane, 2020).

2020; Gutierrez et al., 2022; Bursztyn et al., 2020; Banerjee et al., 2020; Fetzer et al., 2021). We hypothesize that the media may have reinforced emotional responses to shocks by giving greater prominence to Covid-19 issues and, as a result, altering risk perceptions. This mechanism echoes Shachat et al. (2021), which observe a significant increase in risk aversion relative to gains among the study population immediately after two newsworthy events (the announcement of the Wuhan city lockdown and the death of media figure Dr. Li Wenliang). Our results suggest that the source of the media used matters: we do find an exacerbation only when the individual uses social networks.

The remainder of this paper proceeds as follows. In Section 2, we provide an overview of the expected impact of Covid-19 on preferences, and of the mechanisms underlying this potential impact. Section 3 describes our data and preference measurements. Section 4 presents the methodology used to estimate the expected change in attitudes after the onset of the Covid-19 pandemic. Section 5 provides the results and gives some robustness checks. Section 6 discusses the potential limitations of our study, and Section 7 concludes.

## 2 Expected impacts of the Covid-19 crisis on preferences

**Unstable preferences.** There is no consensus on the empirical concept of stability, so we first explain what we mean by instability. The strict definition of the preference stability assumption of Stigler and Becker (1977) empirically implies that an individual's observed preferences must be the same when measured at different points in time. Thus, according to the canonical approach, if we observe a change in preferences, it reflects only measurement errors and should be considered as meaningless noise (Schildberg-Hörisch, 2018). However, given the growing empirical evidence that preferences change significantly over time (due to age or shocks), it seems unlikely that notable changes in preferences are due solely to measurement errors. To harmonize the different concepts of preference instability, Schildberg-Hörisch (2018) proposes a conceptual framework where risk preferences are no longer considered as a single parameter but as a distribution, characterized by its mean and variance.<sup>2</sup> According to this framework, there are three possible reasons for the instability of individual preferences over time. First, it could result from a continuous change in the average level of risk preferences over the life cycle (the aging effect). An observed change

<sup>&</sup>lt;sup>2</sup>The conceptual framework of Schildberg-Hörisch (2018) draws on research in psychology on the stability of individual-specific personality traits (which might be related to what economists refer to as individual-specific parameters determining the curvature of the utility function). Personality traits are considered stable, but should be conceptualized as density distributions that can vary for the same individual in response to a change in their environment. This intra-individual variability is referred to as conditional stability in Fleeson (2001).

in preferences over time may indeed simply reflect the aging of an individual, who becomes less risk-averse and more patient over the life cycle (Yesuf and Bluffstone, 2009; Tymula et al., 2013; Dohmen et al., 2017). Second, some exogenous shocks could induce an abrupt (and potentially permanent) change in the average level of risk preferences. Finally, temporary variations could be observed due to stress or emotions, causing high variance around the average level of preferences.<sup>3</sup> Our analysis excludes the aging explanation because the short period between our three surveys does not capture any life-cycle effect. However, preferences may vary within the time frame of our study because the consequences of Covid-19 are similar to those of a negative shock or because the period of the first half of 2020 was characterized by a high level of uncertainty, generating stress and anxiety (Fetzer et al., 2021).

**Previous literature** Much of the empirical literature studies the impact of natural disasters and conflict, and the evidence is ambiguous. Some studies demonstrate higher levels of risk aversion after exposure to shock (Fang et al., 2022; Cassar et al., 2017; Cameron and Shah, 2015; Reynaud and Aubert, 2020). Conversely, others show a decrease in risk aversion (Biener and Landmann, 2023; Castillo and Hernandez, 2022; Eckel et al., 2009; Kahsay and Osberghaus, 2016; Voors et al., 2012), and some find different results depending on the domain (loss or gain) studied (Reynaud and Aubert, 2020). Regarding time preferences, results are also divergent, as some find that natural shocks increase impatience (Bchir et al., 2013; Cassar et al., 2017) while others show that they decrease impatience (Callen, 2015; Voors et al., 2012). Sample sizes, time horizons, and methodology differ widely across studies and may explain these divergent results. Regarding idiosyncratic shocks such as changes in income, unemployment, health status, or family composition, they do not appear to have an impact on preference changes over time (Chuang and Schechter, 2015; Sakha, 2019). On the contrary, macroeconomic shocks tend to have long-term effects on an individual's risk attitudes (Sakha, 2019). A more comprehensive review of the literature (including life-cycle effects) is available in Chuang and Schechter (2015). These studies are not easily comparable with our, since most of them only observe preferences after the shock occurrence, introducing noise into the causal identification. Moreover, selective attrition (if correlated with temporal stability of preferences) also makes it challenging to construct a valid counterfactual. Within the last years, though, several studies used stronger identification strategies. Among the studies most comparable to our, Hanaoka et al. (2018) track the short-term changes in risk preferences of the same individuals just before and several months after the 2011 Great East Japan Earthquake. They found that

<sup>&</sup>lt;sup>3</sup>Measurement errors are excluded from the framework of Schildberg-Hörisch (2018) because the observed change in individual preferences in empirical studies is systematic and substantial.

women become more risk averse at the high-intensity locations, although they recognize that their results for women are not very robust. Sakha (2019) examine macro- and micro-level influences on changes in risk preference over a period of five years in rural Thailand. They observe high levels of risk aversion during the 2007/08 global financial crisis, following by a phase of return and "normalization" of levels of risk aversion thereafter. More recently, Biener and Landmann (2023) analyze the very short-term (10 days) causal impact of exposure to typhoon in the Philippines on locus of control, beliefs in reciprocity, and risk preferences. They found that people report to be less risk averse immediately after the storm.

The use of natural disasters and macroeconomic shocks to assess the change in preferences after the shock experience, as has been done in the studies cited above, suffers from a significant flaw: their anticipatory nature. This creates the possibility of selective migration and bias in the analysis. In contrast, sudden pandemics, as it was the case with Covid-19, cannot be anticipated and provide a more exogenous and robust framework for identification. Recently, several articles have studied the impact of epidemics. For instance, Fang et al. (2022) analyze the very long term impact of exposure to the 1967 meningitis epidemic in China. They find that intense exposure to the meningitis epidemic during childhood led to increased risk aversion. Regarding the impact of Covid-19, Castillo and Hernandez (2022) and Aragon et al. (2022) are the closest articles to this paper.<sup>4</sup>

<sup>&</sup>lt;sup>4</sup>Various articles and working papers studied the impact of Covid-19 on preferences. These articles are less comparable to ours since they use a different population (students, internet users), in industrialized countries, and often use very short-term time horizons to measure the instability. Among this literature, we can cite Zhang and Palma (2022) that study the short-term (up to 20 days after the emergency declaration) stability of preferences using an experimental design on a US online platform. They find that individuals, on certain risk games, are more risk averse after the pandemic. Angrisani et al. (2020) analyzes the risk preferences of professional traders (48 subjects) and undergraduate students (60 subjects) in London, England, and finds no change in risk aversion levels before and during the Covid-19 outbreak. Drichoutis and Nayga (2022) also find no significant difference in the preferences of 300 students in Athens, Greece, between before and after the pandemic. Gassmann et al. (2022) analyze the impact of the lockdown on time, risk, and ambiguity aversion during the first pandemic wave in France and compare it to what was observed in 2016 (unbalanced panel of 314 subjects aged 18 to 25). They observe a decrease in patience and risk aversion during the lockdown, although the effects tend to disappear after a few months. Bu et al. (2020) exploit a survey of risk-taking behavior among 225 students in the city of Wuhan. They find that people in the hardest-hit provinces of China are even more risk-averse. Shachat et al. (2021) also find a short-term "disruption" in economic preferences from late January to early March 2020 among 396 students in Wuhan. Using online experiments, they provide evidence that people became more risk-averse and impatient than before the Covid-19 pandemic in China. Ikeda et al. (2020) administered an Internet survey in Japan (737 subjects over three waves) and find that people become more risk-tolerant. Harrison et al. (2022) find an increase in atemporal risk aversion pre-post pandemic, while temporal preferences remained stable on an unbalanced sample of 598 online respondents over 6 different waves from a population of undergraduates at Georgia State University (GSU). These studies, however, focus primarily on

Castillo and Hernandez (2022) conducted two survey rounds at the end of 2019 and in May–June 2020 among agricultural smallholders in Guatemela. They observe a significant increase in risk tolerance. Aragon et al. (2022) use longitudinal dataset from a survey of cab drivers in Lima (Peru) conducted before and nearly a year after the pandemic onset. They also find a significant increase in risk tolerance and patience. Beyond the different methodologies and populations studied, the various mechanisms at work in instability may explain why the literature is inconclusive.

**Mechanisms.** Two main mechanisms influence preference instability over time: shock-induced variation in wealth and background risk, and the influence of emotions.

A change in wealth is the mechanism most often cited in the literature to explain the instability of preferences following shocks. According to the habit persistence model, risk aversion varies with wealth level (Campbell and Cochrane, 1999). In this model, increasing wealth leads to a decrease in risk aversion. Cumulative prospect theory also adds that preferences depend on a reference point, not on absolute levels of wealth. This reference point may vary over time and across domains (gains or losses). By causing an unprecedented change in people's environment, the Covid-19 shock may have changed this reference point, modifying preferences accordingly. Because of its unprecedented nature, the Covid-19 crisis can also be viewed as a new background risk, or in other words, a new source of exogenous and non-insurable (economic and health) risks. With the deterioration of the individuals' wealth, Covid-19 should induce greater risk aversion in the economic and financial sphere. However, the change in attitudes may translate to all domains, even those not affected by the Covid-19 shock (Eeckhoudt et al., 1996; Guiso and Paiella, 2008; Gollier and Pratt, 1996). Eeckhoudt et al. (1996) models the fact that individuals become more risk-averse when the background risk becomes riskier, even in other independent risks. This property is described as "risk vulnerability" by Gollier and Pratt (1996). It implies that, in response to this additional background risk and even though the risks are independent, individuals behave in a more risk-averse manner for avoidable risks (Gollier and Pratt, 1996; Eeckhoudt et al., 1996). Individuals reduce their exposure to avoidable risks in order to control their overall risk exposure (Zhang and Palma, 2022; Kimball, 1993). Macroeconomic and financial evidence confirms these theoretical predictions: the presence of (uninsurable) labor income risks alters saving and consumption behaviors, as well as the demand for insurance (Guiso and Paiella, 2008). Therefore, individuals facing new and multidimensional risks should be more risk-averse for other (insurable) risks<sup>5</sup>.

samples that belong to particular demographic groups (such as university students, or a specific profession), which limits the external validity of their results.

<sup>&</sup>lt;sup>5</sup>Kahneman and Tversky (1979) indicate a decreasing sensitivity effect: when the level of risk is already high, adding small independent risks does not change people's behavior. However, it is impossible to estimate the level of

The Covid-19 crisis can also affect individuals' willingness to take risks and impatience by altering their emotions, even if they have not experienced any loss (Guiso et al., 2018; Barberis et al., 2001). For example, Li et al. (2020) and Lerner and Keltner (2001) mention that the fear of contamination influenced risk attitudes toward less risky decisions. Dalton et al. (2020) confirm these findings in a lab-in-the-field experiment with small retail business owners in Vietnam. They report that, consistent with recent lab experiments on risk-taking under stress, small business owners exogenously exposed to financial concerns report higher stress levels and have lower risk aversion than those who received a placebo treatment. Variations in impatience may also be explained by higher emotional stress that tends to reduce self-control (Tice et al., 2001). These studies echo previous literature emphasizing that fear, helplessness, and loss of control, the primary emotional responses to adverse shocks, can alter risk and time preferences (Fang et al., 2022; Lerner and Keltner, 2001; Loewenstein, 2000; Loewenstein and Lerner, 2003; Botzen et al., 2015; Eckel et al., 2009). The emotional pathway might, in some cases, perform better than other explanations: Guiso et al. (2018) show that changes in risk aversion after the 2008 global financial crisis are primarily triggered by fear and a change in perceived probability (while changes in wealth or expected income have no impact). Stress or fear are temporary and, therefore, may induce only short-term changes in preferences. In the same vein, Hetschko and Preuss (2020) show that risk aversion increases after a job loss. Interestingly, this effect is not due to the immediate loss of income but to lower expectations of future income and greater uncertainty about future income. When the emotional response induces a shift in risk perception or discount rate, the behavioral change may persist over time (Ho et al., 2008; Lerner et al., 2015; Brown et al., 2018)<sup>6</sup>. Indeed, shocks could change the perceived utility of bad outcomes and increase the curvature of the utility function (Loewenstein and Lerner, 2003; Guiso et al., 2018; Cassar et al., 2017). It can also affect the expected distribution of returns by altering the salience of some realizations: the salience theory of Bordalo et al. (2012) argues that individuals' attention is in fact directed toward particular realizations that, as a result, receive disproportionate weight. There is no doubt that in 2020, greater importance has been given on the Covid-19 context: most media around the world have crystallized their editorial lines on the news of the pandemic, with some even talking about "infodemic" (Zarocostas, 2020). The higher salience in Covid-19 issues probably intensified the perception of health and economic risks and, in

risk in the pre-Covid-19 Burkinabe environment, or even to estimate the importance of the additional risks induced by Covid-19 relatively to other risks.

<sup>&</sup>lt;sup>6</sup>Some studies show that risk perceptions tend to be strongly altered after exposure to an adverse event. For example, fear causes people to express more pessimistic risk perceptions and leads them to make more risk averse choices. Brown et al. (2018) also find a change in risk perceptions, as well as in the individual's beliefs about the frequency and magnitude of future shocks

this way, may have influenced risk and time preferences.

## **3** Data and Descriptive Statistics

#### 3.1 Panel Survey of Women in Ouagadougou

**Baseline survey.** We rely on an original panel survey that is part of a larger project aiming to understand the health and economic behaviors of women working in the informal sector in Ouagadougou, the capital of Burkina Faso.<sup>7</sup> We conducted a baseline survey in January and February 2020 (face-to-face interviews, see Table B1). At that time, no respondent could have anticipated the coronavirus crisis in Burkina Faso or its economic consequences. Our sample comprised 1,700 female loan recipients randomly selected from two microfinance agencies. In this baseline survey, we collected information on the socioeconomic characteristics of respondents and their household members and their health and work behaviors.<sup>8</sup> We also asked questions about hypothetical lottery choices to elicit individual risk and time preferences (see the following subsections).

Midline survey. In June and July 2020, we conducted a follow-up telephone survey. Due to financial and logistical constraints, only half (randomly selected) of the baseline sample was reinterviewed. Before randomly selecting participants for follow-up, we excluded women without cell phones (1% of baseline survey respondents) from our sampling frame. Our follow-up rate was exceptionally high, reaching 95% of randomly selected participants.<sup>9</sup> Although selective attrition is not an issue for this round, we can confirm in the table that there are no significant differences between respondents selected to participate in the midline survey and those who are not. The balancing tests between these two types of respondents are reassuring (Table B2 in Appendix),

<sup>&</sup>lt;sup>7</sup>This project is being conducted in partnership with a local microfinance institution with several agencies in Ouagadougou. The main objective of this large project is to assess the impact of a novel health-insurance program, introduced in 2021, on a range of behaviors, including health, work, women's empowerment, and risk and time preferences. The first two waves of surveys were conducted several months before implementing the health-insurance program.

<sup>&</sup>lt;sup>8</sup>Note that our measure of wealth refers only to living conditions (location of water supply, type of toilet, electricity, type of fuel used for cooking and lighting) and asset ownership (radio, television, cell phone, refrigerator, car, motorcycle, farmland). Indeed, income and earnings are inappropriate for informal workers because they are very volatile. In addition, it is difficult to compare women's income levels because some pool their income with their husbands, but others do not. In addition, in the pilot survey, we found that these questions about individual or household income and earnings were embarrassing because they made respondents uncomfortable.

<sup>&</sup>lt;sup>9</sup>The telephone survey protocol required interviewers to make three contact attempts in two days, with at least one day between attempts. We did not provide financial incentives to follow-up respondents.

although we find that midline sample respondents are significantly more likely to have a formal job (still less than 5%) and more likely to own a television. Other characteristics (e.g., health status, employment status, financial status, preferences) are not significantly different between the respondents selected for the baseline and midline and those interviewed only in the baseline. In this follow-up survey, we re-asked questions about respondents' health and employment status and their risk and time preferences to observe these characteristics at two different time points. Socioe-conomic characteristics (such as education or standard of living as measured by assets owned) were only collected in the first wave. Indeed, we assumed that these characteristics were unlikely to have changed over the six months. These variables are therefore considered invariant for the 2020 period, or in other words, not affected by the Covid-19 crisis in the short-term. However, we supplemented the follow-up questionnaire with a series of questions on perceptions of the coronavirus, ways to obtain information, and specific economic impacts of the Covid-19 crisis.

Endline survey. We conducted the endline survey in January/February 2022, in person, so that the conditions would be as close as possible as the ones of the baseline survey (same season, same training). In a consistency perspective, the field team (all managers and three quarters of the surveyors) remained the same. The attrition rate between baseline and endline is very high (57.98%) and come from individuals having left the microcredit agency. Table B3 in the Appendix shows us that, except for risk and time preference levels, there are significant differences between attritors and non-attritors on several characteristics, such as the sector and the location of the economic activity, or the probability of having been sick in the three months prior to the survey. Although significant, the differences in characteristics are small in their magnitude and should not affect the results. As our analysis intends to track the evolution of preferences over time for the same individual, we do not correct selective attrition.

#### 3.2 The COVID-19 crisis in Burkina Faso

Government response to COVID-19. As in most African countries, Burkina Faso recorded few cases of Covid (at the end of our follow-up survey in July 2020, the number of confirmed cases in Burkina Faso was 1,000, and the number of deaths was 54). These low numbers are partly due to strict measures restricting the movement of people, goods, and services implemented from mid-March 2020. Those restrictions included a two-and-a-half-month curfew, a ban on public gatherings and events, the quarantine of major cities, the closure of schools, bars, restaurants, public markets, and religious places for several weeks and the disruption of public transportation for several months. Each week, the government made a public communication to inform the population about the evolution of the pandemic and the new measures put in place. In addition, prevention messages were widely disseminated through public and social media, such as wearing masks, washing hands with soap, using alcohol-based hand sanitizers, and social distancing.<sup>10</sup> Restrictions had been withdrawn by the time of our follow-up investigation, although protective measures such as social distancing are still strongly recommended.

**Economic impacts.** The low number of infections raises the question of the real impact of the Covid-19 epidemic on the daily lives of Burkinabes, particularly a few weeks after the measures were relaxed. First, the population of Ouagadougou did comply with the measures put in place (Mathonnat et al., 2021). The figure C1 in Appendix C illustrates population movements in the Ouagadougou region based on Google community mobility trends. It shows a substantial increase in the location of private residences and a sharp decrease in mobility to recreation, grocery stores, and workplaces over our study period (in blue). In addition, our data suggest that respondents were aware of the measures in place. First, 98% cited public places as the most significant risk of contamination. The majority of people agreed (40%) or strongly agreed (54%) with the economic measures taken by the government. In addition, 88% of those surveyed in June and July 2020 still wore masks, while only 54% respected the social distance. The likely trade-off between the need for sustainable income generation and compliance with government measures may explain why recommended sanitary practices were not fully respected. These results are consistent with those found in Mathemat et al. (2021), where respondents report compliance with sanitary and social distancing habits in general but admit to not following them all. Despite the imperfect compliance and the period of return to normalcy during the second wave, the restrictive measures adopted had negative consequences that persisted in July 2020. First, we see an increase in concerns about health and economic issues. For example, 35% of respondents anticipated a worsening economic situation in the medium term (Table C7 in Appendix). Second, the economic situation deteriorated between the two survey waves: only 83% of respondents were working the week before the second interview, whereas this was the case for 92% of them in the first wave. In addition, 54% of respondents reported frequent restrictions in their economic activities, and 7% said that these restrictions were systematic (Table C6 in Appendix). These statistics echo those found in studies describing that vulnerable (informal) workers in Africa have been disproportionately affected by the Covid-19 crisis

<sup>&</sup>lt;sup>10</sup>Because no official website has centralized all actions taken or recommended, we rely on a review of major online newspapers, the government website, and social networks (Facebook). For a timeline of government responses to the real-time assessment of the epidemic situation, see https://askabout.io/covid-19/ask/what-is-the-government-response-timeline-for-burkina-faso/.

#### (Price, 2020; Mahmud and Riley, 2020).

#### 3.3 Measuring risk and time attitudes

**Risk attitudes.** To obtain information about attitudes toward risk, we used non-incentivized hypothetical gambling questions similar to Holt-Laury Paired Lottery Task (Holt and Laury, 2002). Each respondent was presented with three scenarios involving paired lotteries in an abstract environment with payoffs ranging from 0 to 20,000 FCFA. For each scenario, the respondent was asked to choose between an option where the payoff is certain (option A) and an option with two possible payoffs with equal probabilities (option B). The first pair of choices gives a higher expected payoff for the riskier choice (option B). On the contrary, the last pair of choices gives the same expected payoff for the option B as the option A (the safer choice, see Figure A1). We replicate the same experiment in the loss domain (Figure A2). Tables A1 and A2 illustrate how we measure risk aversion in the gain and loss domains. The number of risky decisions made by the respondent measures the individual's degree of risk aversion in each domain on a scale of 1 to 4. Specifically, we construct ordinal variables ranging from 0 (high risk taking, i.e. consistently choosing the lottery) to 3 (high risk aversion, i.e. consistently choosing the safe option). In the main empirical analyses, we use a binary variable indicating that the respondent is "risk averse", i.e., always chose the safer option.

Figure 1 below shows how risk and time attitudes changed between the two (balanced panel on baseline and midline) or the three surveys (balanced panel on baseline, midline and endline). While the sample size differs, a similar trend can be observed. In the gain domain, individuals tend to be risk averse: in January 2020, 75% (/72%) of respondents from the two-waves panel (/the three-wave panel) were in the most risk averse category (see also Table A4). The Covid-19 crisis has increased risk aversion, with this proportion rising to 83% (86%) in June 2020. For respondents also observed in January 2022, the proportion of risk-averse individuals has returned to about its initial level, at 74% precisely. Consistent with the reflection effect of Kahneman and Tversky (1979), only 60% (/61%) of people from the two-waves panel (/the three-wave panel) showed high risk aversion in January in the loss domain (Table A5). This proportion dropped to 30% (/28%) in June 2020. Interestingly, the proportion of risk-averse individuals was only 37% in January 2022, suggesting a long-term effect of Covid-19 on risk aversion in the loss domain.

**Time preferences.** Following Andersen et al. (2008) and Cassar et al. (2017), time preferences are obtained using four sets of choices. Each set consists of a choice between hypothetically receiving 5,000 FCFA on the day of the survey or hypothetically receiving 20,000 FCFA in a longer time frame. We set up four different time horizons that vary according to the choice set: the day of the

survey versus one month, three months, six months, and twelve months (Figure C3). To measure subjects' impatience, we create a categorical variable ranging from 0 (High patience) to 4 (High impatience). We also use in the empirical analysis a binary variable indicating that the respondent is "impatient", capturing by the last category. We observe in Figure 1 an increase in preference for the present after the Covid-19 crisis: individuals from the two-waves panel (/the three-wave panel) were more impatient in June 2020 (79% / 83%) compared to January 2020 (65% / 68%). For the sample of individuals observed at 3 points in time, a high degree of impatience remains in January 2022, at 76% (Table A6). The following section extends the analysis to examine whether these changes hold once we control for observable and unobservable characteristics.

Validation of preference measure We build our preference measures on hypothetical and nonfinancially incentivized questions. Several studies have demonstrated the ability of these survey methods to capture individuals' induced preferences and accurately predict the attitudes measured in a more experimental setting or in real life (Fang et al., 2022; Dohmen et al., 2011; Guiso et al., 2018). However, various biases may affect the validity of these methods in capturing Burkinabes' risk and time preferences. First, understanding the questions may be challenged when addressed to a population with limited education. To limit these comprehension biases, we presented each scenario with a visual representation of the amount of money the respondent could earn by choosing options A and B (see for example Figure A4). In addition, before each set of risk questions, we checked that the person understood the game by asking them to choose between option A (e.g., for the risk in the gain domain, having a certain gain of 5000 FCFA) and a non-rational option B (a 50-50 chance of winning 5000 FCFA or nothing, see Figure A5). Few people chose the nonrational option in these pre-game questions (respectively, 5% for the gain domain, 6.8% for the loss domain). If they did, the interviewers re-explained the game. Another concern relates to the calibration of the game amounts or time horizon, since unsuitable scenarios can induce noise in the measurements and affect the understanding of the games. On the one hand, we extensively tested the calibration of the amounts and time horizon in the pilot phase of the survey (in November 2019) and solicited feedback from pilot project participants. On the other hand, we can see from Figure 1 that preferences are quite polarized, since few individuals fall into the intermediate categories. Indeed, we observe little variation in the options changes for the same individual in the same wave of survey. More precisely, 95% of the individuals who chose option A in the first scenario will stick with this option in the last scenario of the game (and similarly if the person chooses option B in the first scenario). Regarding the risk measures, we also tested whether a tipping point appeared by adding a scenario, namely the choice between +/-5000 euros for sure and a lottery of +/-25000 euros. The addition of this extra scenario keeps the risk attitude categorization the same.

Concerning time preferences, we also tested a fourth scenario, reducing the time horizon of Option B to one week. Similarly, adding this alternative keeps the choices the same, i.e., respondents who chose option A for a time horizon of one week consistently choose the same option when the time horizon is extended to one month, and vice versa for option B. There is also no change between a time horizon of 6 months and one year. A final calibration test was performed on time preferences by increasing the gain to 40,000 FCFA francs. Again, individuals are not very sensitive to the design of the options and remain consistent with their choice.

**Predictive power of our preference measures** We check whether our preference measures were actually associated with real behaviors regarding risks. In our midline survey, we asked respondents whether they had interrupted their activity for a day or more during the Covid-19 period, and the reasons associated with the interruption (or non-interruption, if any). Responses include options related to restrictions and others pertaining to individual fears about the virus. We consider that stopping an activity for fear of contracting or transmitting the virus reveals greater risk aversion than stopping activities to comply with restrictions, as the latter individuals would have continued to work without legal restrictions. On the contrary, continuing to work despite restrictions is a risk-loving attitude because it implies that people accept the consequences of potential punishment if caught and possible illness if infected. We construct a risk-loving behavior variable, with a value of 0 if the respondent stopped working because they feared contracting or spreading the virus (and did not mention restrictions as a reason), 2 if respondents continued working because they could not afford to have any income (and no other reason), and 1 for different situations. We also know how respondents feel about the country's economic development one year from now (measured on a Likert scale, ranging from strongly worse to strongly better). We argue that this is an immediate and concrete indicator of respondents' willingness to engage in risky ventures (taking out a loan to start or expand a business, for example), at the point where all restrictions have been lifted. We estimate a linear model to test the relationship between these two measures and our measures of risk and time preference. Our results (Table 1) show that our risk aversion measure significantly and negatively predicts risk-loving behaviors. In contrast, risk aversion in the loss domain and present preference have significant and negative coefficients. These results are robust to introducing control variables related to individual characteristics and the frequency with which individuals experienced difficulties seeking care or finding medication during Covid-19. These results, coupled with the literature validating the use of hypothetical lotteries to reveal preferences, support our belief that our preference variables accurately measure respondents' mindsets regarding risk and temporality.

	Risk-loving behavior		Beliefs on economic evolutio		
	(1)	(2)	(3)	(4)	
Risk aversion (loss)	0.001	-0.006	-0.181***	-0.187***	
	(0.013)	(0.032)	(0.013)	(0.033)	
Risk aversion (gain)	-0.037**	0.035**	-0.017	0.017	
	(0.019)	(0.047)	(0.018)	(0.047)	
Preference for present	0.003	0.010	-0.131***	-0.128***	
	(0.016)	(0.041)	(0.016)	(0.042)	
Controls		$\checkmark$		$\checkmark$	
Nb. Observations	730	727	730	727	

Table 1: Validation of the risk and time preference measures

Figure 1: Changes in risk and time preferences between January and June 2020



#### **Risk aversion - Loss Domain**

🚥 0-High risk taking 📁 1-Moderate risk taking 🥅 2-Moderate risk aversion 🥅 3-High risk aversion



#### **High impatience**



## 4 Changes in risk and time preferences

#### 4.1 Empirical specification

To test whether attitudes toward risk and time have changed due to Covid-19, we denote as  $Y_i$  the outcome of interest, representing the different measures of preferences (risk aversion in the gain domain, risk aversion in the loss domain, or impatience). Our treatment variable is a time dummy taking the value of 1 for the post-Covid-19 period (June 2020 and January 2022) and 0 for the pre-Covid period (January 2020).<sup>11</sup> We estimate two types of effects from two different samples. The first is a short-term effect, based on the balanced panel of 853 women interviewed between baseline (January 2020) and midline (June 2020). The second one is a medium-term effect, based on the balanced panel of 390 women interviewed three times, at baseline (January 2020), midline (June 2020), and endline (January 2022). Pooling the several waves, we estimate our model as follows:

$$Y_i = \alpha_i + \beta \cdot PostCovid_i + \theta_i + \delta W_i + \mu Y'_i + \varepsilon_i \tag{1}$$

In order to present a clearer relationship between preferences and the Covid-19 crisis and to control for any spurious associations that might bias our estimates, we include in some specifications individual fixed effects  $\theta_i$  that eliminate any time-invariant characteristics. We also control for  $W_i$ , i.e., having worked in the week prior to the survey. Moreover, as recommended in this literature, we include the time (or risk) preferences  $Y'_i$  to account for relationships between these variables (Dohmen et al., 2010; Bchir et al., 2013; Cassar et al., 2017).

In the short-term,  $\beta$  captures the causal effect of the Covid-19 crisis on preferences. This is the global and multidimensional effect of Covid-19, encompassing changes in income, occupational status, stress, and uncertainty in the first six months of 2020. Our estimates are unlikely to suffer from reverse causality or selection bias because of the exogenous and unexpected nature of this singular shock. Given the importance of this single event, we consider any change during this period in economic (income, employment) and emotional (stress, uncertainty, anxiety) outcomes to be related to Covid-19, either directly or indirectly. However, some omitted variables may remain

<sup>&</sup>lt;sup>11</sup>The use of a dummy variable as a treatment implicitly implies that the Covid-19 crisis affected every woman in our sample homogeneously. In our case study, this is not a strong assumption: as mentioned earlier, our sample is composed of women microentrepreneurs living in the same area (Ouagadougou) and exposed to the same implemented measures and health risks.

and lead us to misinterpret any change in preferences over the analysis period as being caused solely by Covid-19. Some factors, time-invariant, are captured by the individual fixed effects. For example, the willingness to take long-term risks or the curvature of the individual utility function and its risk premium should be considered as personality traits, which vary little over the life cycle. Other determinants of preference instability, such as education or age, are unlikely to have changed over the 6-month period. These characteristics are also absorbed by fixed effects. An exception may be changes in wealth over the period that may have induced a change in attitudes, if they are reference-dependent. As explained earlier, our empirical measure of wealth focuses on asset ownership to avoid the reporting bias associated with income. Indeed, income emerged as a sensitive issue in the pilot survey and was difficult to compare across individuals, since some women do not pool their income with family members. Asset ownership is a very stable measure of living standards, and it is unlikely to have changed over the study period for reasons non-related to the Covid-19 crisis. However, a change in income between January and June 2020 unrelated to Covid-19 may have happened. We believe that if unrelated to Covid-19, i.e., not related to existing restrictions or declining overall economic activity, this short-term income change should not alter preferences. Indeed, seasonality in income is predictable and to our knowledge, no study has shown that attitudes toward risk and time change with seasonal fluctuations in income. In addition, no other shocks occurred in Ouagadougou between January and June 2020.

Over a two-year horizon, we are able to complement our analysis by tracking the persistence of shifts in respondents' preferences over a longer time. The significance of coefficient  $\beta$  indicates whether the changes observed over the first semester 2020 were temporary because preferences are back to their initial level (if  $\beta$  is non-significant) or they persisted over time are still observed 2 years after (if  $\beta$  is non-significant). Nonetheless, we acknowledge that one needs caution while attributing the causal effect of Covid-19 to coefficient  $\beta$ , for several reasons. First, the individual fixed effects are no longer sufficient since, during these two years, individuals may have increased their level of education and changed their occupation, family situation, and wealth status. Therefore, we control for the characteristics likely to vary during these two years and are well-known predictors of risk and time preferences. Second, between January 2020 and January 2022, Burkina Faso experienced several significant events, such as the continuous deterioration of the national security situation, the food crisis, high inflation (the inflation rate was 17.8 in June 2022), and the coup of January 2022. It is, therefore, difficult to distinguish the effect on preferences of the persistence of the Covid shock from that of the other shocks that have passed through Burkina Faso in two years and to make a safe causal inference.

Another possible explanation for the observed variation in preferences over time is related to measurement issues. We explained in the previous section that our measures were calibrated to best fit the context and designed to avoid misunderstandings with respondents as much as possible. In addition, cognitive ability, education, and age, which are strongly correlated with noisy behavior in risk elicitation experiments (Andersson et al., 2020; Choi et al., 2014), are captured by individual fixed effects. Thus, we assume homogeneous noise (i.e., classical measurement errors) over time for each individual in our estimates. In other words, once fixed effects have excluded cognitive abilities, arbitrary and inconsistent responses can be considered random. Another concern relates to the mode of administration of the surveys: for the midline wave, surveys were conducted by phone. Phone surveys exclude visual aids and may alter the understanding of the game. We argue that this is unlikely to happen since the baseline and midline surveys are very close in time, so the respondent faced the same decisions at two points close in time. The short timespan between surveys actually increases the risk of path dependence, i.e., perfect correlation over time for each individual. As a result, the probability of finding variation in preferences is reduced. In addition, in the endline we went back to a face-to-face survey (with the same interviewers) which is, therefore, comparable to the baseline. Lastly, one may be concerned that in the midline, the seasonality is different, which may explain the variability in preferences. While this concern may apply to short-term effects, it is no longer relevant to long-term effects as the baseline and endline surveys are conducted over the exact same period (January and early February). Although we cannot rule out the existence of some measurement error in our estimates, it seems unlikely that noise alone could cause such large differences in individual preferences.

#### 4.2 Main results

Changing attitudes towards risk in the gain domain. Table 2 presents estimates of the impact of the Covid-19 crisis on risk attitudes in the gain domain, using linear probability models. The dependent variable is binary, representing high risk aversion. All specifications allow for robust standard errors. We separate short-term estimates, based on the balanced sample between the baseline and the midline, and medium-term estimates, based on a balanced sample of the three survey waves.

We begin with column (1) and (4), which estimates the short-term and medium-term impact of Covid-19 on risk aversion without any controls or fixed effects. Consistent with what we observe in Figure 1, the estimates confirm a significant shift in risk attitudes in the gain domain towards

	Short-term variation			Mediu	Medium-term variation		
	(1)	(2)	(3)	(4)	(5)	(6)	
Post	0.090***	0.092***	0.093***	0.078***	0.079***	0.069**	
	(0.019)	(0.019)	(0.020)	(0.027)	(0.027)	(0.029)	
Baseline mean $(\%)$	74.53			71.90			
Relative effect $(\%)$	12.03	12.08	12.08	10.85	11.13	9.74	
Nb. Observations	1710	1710	1604	1087	1087	1043	
Robust standard errors	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
Individual fixed-effect		$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$	
Controls			$\checkmark$			$\checkmark$	
Nb. Observations	1710	1710	1604	1087	1087	1043	

Table 2: Temporal variation of risk aversion (gain domain)

Notes: Short-term variation : on balanced panel between baseline and midline.

Medium-term variation: on balanced panel between baseline, midline and endline.

*Post* is a dummy variable indicating the second wave and the third wave.

Controls include having a job the week before the survey, age, level of education, asset score, household size, subjective health status.

Robust standard errors in parentheses. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

greater risk aversion after the Covid-19 crisis, both in the short and the long-term. On average, individuals are eight percentage points more likely to choose the safest option. For ease of interpretation, we calculate the relative effect of each estimate using the average level of risk aversion of January 2020 as a base reference. In terms of relative effects, our results show a 12% increase in risk aversion between January and June 2020 (column (1)) and a 11% increase between January 2020 and January 2022. Using individual fixed-effects (column (2) and column (5)) does not change the magnitude of the effect. This implies that the instability in risk aversion associated with Covid-19 in the gain domain is not caused by time-invariant characteristics, such as underlying fundamental risk preferences, cognitive abilities, education, age, or personality traits. In the short-term, the estimates are also remarkably stable when we control for being employed in the week before the survey (column (3)). This suggests that job instability during this period does not explain the variation of preferences. The relative contribution of the individual level of impatience (a dummy taking the

value of 1 if the individual has a strong preference for the present) is marginal, as the average effect is barely affected when it is included. In the medium-term regressions, the relative effect is slightly smaller than in the short-term ones, especially when controls are taken into account. This suggests that individual experiences such as job loss, wealth, or health status have only a marginal impact on preference changes over a longer timescale.

Table D1 in Appendix D presents several robustness checks using the ordinal variable instead of the binary one. The results are about the same, which is not surprising given the highly polarized reactions to the games. Table D2 present alternative specifications. Precisely, column (1) and column (4) replace the robust standard errors with bootstrapped standards errors, leaving the results unchanged from the main specification, both for the short and the medium-term. Columns (2) and (5) present the results of the estimation using random effects. Unlike fixed-effects estimates that absorbed time-invariant characteristics, in this specification, we control for age, education, employment status, standard of living, number of people living in the household, and subjective health status. The specification with random effects leads to a similar conclusion: the Covid-19 crisis increased individual risk aversion in the gain domain. We also test whether our results can be influenced by individuals suffering from chronic stress and anxiety by excluding individuals who reported suffering from it in the baseline survey (5% of the initial sample). The results barely change when we restrict the sample to non-anxious individuals (column (3) and column (6)).

**Changing attitudes toward risk in the loss domain.** One of the contributions of our study to the preference instability literature is to provide evidence on how shocks can influence risk preferences in the loss domain, in the same vein as Reynaud and Aubert (2020). The impact of Covid-19 may be different depending on the domain under study: an individual is more likely to be risk-averse in the domain of gain but risk-lover in the domain of loss (Kahneman and Tversky, 1979; Levin et al., 2012; Reynaud and Aubert, 2020).<sup>12</sup> Little literature provides clues to explain these differential behaviors. The psychological literature mentions a selective allocation of attention to loss or gain outcomes: people tend to pay more attention to outcomes framed as losses than presented as gains (Willemsen et al., 2011; Pachur et al., 2018). Table 3 shows the estimates when the lotteries are expressed in terms of losses. We find different responses to the Covid-19 crisis

<sup>&</sup>lt;sup>12</sup>The reflection effect has been demonstrated in prospect theory (Kahneman and Tversky, 1979; 2013). People feel losses more intensely than gains. They thus prefer to choose riskier lotteries in the loss domain, offering a possibility of a minimal loss rather than a certain loss. The attitude towards risk is reversed in the gain domain, where the certain gain is preferred in most cases (Li et al., 2011). Some empirical studies show that the error term is smaller in the gain domain than in the loss domain (Lopes, 1987; Gonzalez et al., 2005; Baucells and Villasís, 2010), suggesting that choices in the loss domain create more conflict and cognitive effort and are therefore less stable.

in this domain: the magnitude of the effect is much more substantial in the loss domain than in the gain domain, and the attitude toward risk is reversed. Indeed, we find that respondents are more likely to choose risky options after the Covid-19 crisis. The results show that, on average, individuals are 38 percentage points less likely to choose the safest option. In other words, we observe a 47% decrease in risk aversion in the loss domain between January and June 2020. After two years, we continue to see a decline in the propensity to take risks, of the same magnitude. Risk aversion also decreased by 47% between January 2020 and January 2022, suggesting a permanent Covid-19 effect. Table D1 and Table D2 reproduces the same robustness checks explained in the previous subsection. We find that the results are stable in terms of magnitude regardless of the definition of the dependent variable (categorical instead of binary) or the specification used (bootstrapped standard errors, random fixed effects with time-invariant controls). These results indicate that individuals are more likely to take risks in the loss domain following the Covid-19 crisis. This result is similar to those found in Laury and Holt (2008) and Shachat et al. (2021).

	Short-term variation			Medium-term variation			
	(1)	(2)	(3)	(4)	(5)	(6)	
Post	-0.297***	-0.296***	-0.281***	-0.286***	-0.288***	-0.288***	
	(0.022)	(0.023)	(0.026)	(0.030)	(0.030)	(0.032)	
Baseline mean (%)	60.12			61.13			
Relative effect $(\%)$	-49.46	-49.9	-46.57	-46.76	-47.44	-47.44	
Robust standard errors	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
Individual fixed-effect		$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$	
Controls			$\checkmark$			$\checkmark$	
Nb. Observations	1701	1701	1596	1083	1083	1039	

Table 3: Temporal variation of risk aversion (loss domain)

Notes: Short-term variation : on balanced panel between baseline and midline.

Medium-term variation: on balanced panel between baseline, midline and endline.

Post is a dummy variable indicating the second wave and the third wave.

Controls include having a job the week before the survey, age, level of education, asset score, household size, subjective health status.

Robust standard errors in parentheses. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

While being a nice illustration of the reflection effect of Kahneman and Tversky (1979), the opposite direction of impact may concretely illustrate that different dimensions of Covid-19 crisis trigger each domain. In a recent study, Galandra et al. (2020) show a preferential allocation of attentional resources to the most relevant domain during the Covid-19 outbreak in Italy. Specifically, the authors use a modified version of the Holt-Laury Paired Lottery Task that is explicitly linked to adverse health and employment outcomes in the context of the Covid-19 pandemic. They observe risk-averse behavior with employment-related lotteries and risk-seeking behavior with health-related lotteries. They suggest that individuals may be more willing to make risky decisions for their health to achieve the best possible outcome. However, people are concerned about the potential long-term economic consequences, for instance in their employment conditions, so they exhibit greater risk aversion in this dimension. We can easily extend the conclusion of this suggestive evidence to consider that time preferences also respond to different aspects of this multidimensional crisis.

**Change in impatience** Table 4 presents the results for the impact of Covid-19 on impatience. This table uses the same specifications as those used for risk aversion, except that we control for risk aversion in the gain domain (in columns (3) and (6)). We observe that respondents become 13 percentage points more impatient just after the Covid-19 crisis (columns (1) to (3)). We also provide relative effects to assess the magnitude of the impact: impatience increased by 20% between January and June 2020. On the medium run, we also observe a 16% increase in impatience between January 2020 and January 2022. Although slightly lower than immediately after the Covid shock, preferences for the present have not returned to their initial level two years later. Again, as in the case of risk aversion, the results do not vary when we run different specifications or use different definitions (see Table D1 and Table D1). The observed increase in impatience due to Covid-19 is consistent with the results obtained by Bchir et al. (2013) and Cassar et al. (2017).

#### 4.3 Our results in the literature on instability of preferences

Our findings fit into a very diverse empirical literature, where the results depend strongly on the individual characteristics of the respondents and the measurement methods (e.g. stated or revealed). Indeed, the lasting decrease in risk aversion that we observe in the loss domain is consistent with the results of Aragon et al. (2022), Biener and Landmann (2023), both of which use a stated and a revealed measure of earnings-oriented risk aversion, and Castillo and Hernandez (2022), which uses only a reported preference measure. In contrast, the increase in risk aversion observed in the gain domain could only be related to the long-term effect found by Fang et al. (2022). Several reasons can be given to explain these differences. The first is population characteristics: both the Aragon

	Short-term variation			Mediu	um-term variation	
	(1)	(2)	(3)	(4)	(5)	(6)
Post	0.132***	0.132***	0.133***	0.116***	0.116***	0.113***
	(0.020)	(0.020)	(0.023)	(0.027)	(0.027)	(0.029)
Baseline mean $(\%)$	64.41			66.94		
Relative effect $(\%)$	20.50	20.18	20.18	17.38	17.93	16.43
Robust standard errors	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Individual fixed-effect		$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$
Controls			$\checkmark$			$\checkmark$
Nb. Observations	1742	1742	1604	1099	1099	1043

Table 4: Temporal variation of high impatience

Notes: Short-term variation : on balanced panel between baseline and midline.

Medium-term variation: on balanced panel between baseline, midline and endline.

Post is a dummy variable indicating the second wave and the third wave.

Controls include having a job the week before the survey, age, level of education, asset score, household size, subjective health status.

Robust standard errors in parentheses. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

et al. (2022) and Castillo and Hernandez (2022) samples are predominantly male. Gender is an important determinant of the level of preferences. If our mostly female sample is initially more risk averse than theirs, as the evidence suggests, this could explain the increased risk aversion in the earnings domain. In contrast, Biener and Landmann (2023) have a predominantly female sample (83% and 79% respectively before and after the shock), but on average, more educated than ours. Given that they find a larger effect on the educated and that our sample is predominantly low-educated, the difference in average results could stem from the fact that our sample concentrates in the low-educated whose risk aversion is most likely to change, because they lack coping mechanisms such as insurance, savings, or labor market opportunities.

Nevertheless, the characteristics of the study population certainly do not explain the discrepancy between the results in the win and loss domains. Since these studies do not use separate lotteries for gains and losses, we can only speculate on the reason for these differences. Kahneman and Tversky (1979) provides a promising element, emphasizing the importance of the reference point in the analysis of any choice. An individual who has suffered a loss but has not adapted to this new situation might include his loss in any proposed lottery and, as a result, consider a gain lottery as a loss lottery. From this perspective, the proposed gain lotteries by Aragon et al. (2022), Biener and Landmann (2023) might have been analyzed by storm or Covid-19 respondents as broader loss lotteries that included what they lost with the shock, hence the correspondence with our loss lottery results. By this logic, post-shock respondents may be more likely to engage in risky strategies to limit their overall losses even when self-assessing their aversion. Given the lack of insurance or even public compensation for private losses in many developing countries after large shocks and the severity of the economic crisis that follows these events, it is reasonable that this time lag exceeds 18 months (the time lag in our study and in Aragon et al. (2022)).

## 5 Identification of potential transmission channels

#### 5.1 Actual impact versus concerns regarding Covid-19 experience

Our main results measure the overall impact of Covid-19 without differentiating which dimensions of this crisis generate a change in preferences. In this section, we attempt to shed light on potential underlying mechanisms behind our findings. The graphs below present the results of interactions between the Post-Covid binary variable and various self-reported Covid-19 impacts for the short-term sample.<sup>13</sup> These measures suffer from reporting bias and should not be interpreted as causal mechanisms. Nevertheless, they allow us to differentiate Covid-19's impact on daily life (actual self-reported impact), which is closer to the economic and health pathways, from concerns induced by the Covid-19 crisis, which are more related to the emotional channel.

Two graphs are presented for each domain (gain domain in Figure 2, loss domain in Figure 3, time preferences in Figure 4). The first graph identifies the mechanisms that actually impacted respondents' daily lives: employment status (having a job dropped from 92% to 83% between the two survey waves), having an average standard of living below the sample median, having regularly ran out of food or water during the Covid-19 period (37% of households). We find few effects of these actual economic impacts: having a job significantly attenuates the decline in risk aversion in the loss domain only. In contrast, running out of food or water significantly increases risk aversion

<sup>&</sup>lt;sup>13</sup>Graphs for the medium-term sample are available upon request, although comments remain the same with this sample.

in the gain domain. We also examine health-related impacts, such as whether individuals experienced coughing and/or breathing difficulties during the period (4.9% of respondents), or whether they experienced Covid-19 symptoms (fever and coughing or breathing difficulties, and vomiting or diarrhea - 2% of respondents). We find no effect, potentially due to the low prevalence of these symptoms. Besides, as noted above, there were very few confirmed cases of Covid-19 at the time of the surveys. Finally, we interact with respondents' self-reported health status (on a scale of 1 to 4) and only with those who reported poor health (15% of respondents), to isolate those most at risk. We find no effect of these variables on the variability of preferences over the period (both for short-term and medium term).

The graphs on the right refer to the respondents' concerns generated by Covid-19. Only 2% of respondents reported having no concerns (see Appendix B). We distinguish several types of concerns: fears that the respondent or a family member will contract the disease (51% of respondents), fears of running out of money (for food, medicine or credit repayment, 33%), fear that the economy will collapse (40%), that it will no longer be possible to travel (42%) and that the disease will spread everywhere (49%). We can see that fears related to the disease have no effect. On the other hand, fears associated with the economic context have a significant effect on risk aversion: being afraid of running out of money increases risk aversion for gains (coefficient of 0.18, corresponding to an additional relative increase of 24%) and decreases risk aversion for losses (coefficient of -0.23, corresponding to an additional relative decrease of 38%). There is also a strong and significant effect of fear of economic collapse (+16% additional increase in risk aversion in the gain domain, and -20% additional decrease in risk aversion in the loss domain). Fear of travel restriction increases risk-taking in the loss domain (-25% additional risk aversion) and increases impatience (+13% additional).

Although prone to reporting bias and measurement errors, it is striking that the actual effects of the Covid-19 crisis had little impact on preferences, while economic concerns significantly increase preference variability over the period. This result suggests that the main channel of preference variation due to Covid-19 is emotional. These results are similar to those of Fetzer et al. (2021) who find an increase in economic anxiety in the early weeks of Covid-19, shaped by individuals' beliefs about the mortality and contagiousness of the coronavirus. In our context, the health risk is less salient than the economic risk, and it is indeed economic concerns that, months after the onset of the crisis and weeks after the end of the restrictive measures, influence preference instability. This result is consistent with the psychosocial literature, which shows that fear is related to risktaking. For example, Lerner and Keltner (2001) show that more fearful individuals are less likely Figure 2: Actual impact and concerns regarding the Covid-19 crisis - Risk aversion - gain domain



Risk aversion - gain domain (Short-term sample)

to take risks in a hypothetical situation. In an experiment involving randomized electric shocks, Cohn et al. (2015) also show that the fear of the electric shock (and not the actual shock) decreases financial risk-taking. Figure 3: Actual impact and concerns regarding the Covid-19 crisis - Risk aversion - loss domain



## Risk aversion - loss domain (Short-term sample)

Figure 4: Actual impact and concerns regarding the Covid-19 crisis - Impatience



# Impatience (Short-term sample)

#### 5.2 Media exposure

Another indirect way to look at the role of emotion is to examine the role of media exposure during the Covid-19 period. Fetzer et al. (2021) indeed, show that news conveying the mortality and contagiousness of Covid-19 increases economic anxiety. In their study, Shachat et al. (2021), Sadish et al. (2021) also illustrate this mechanism as they observe a significant increase in risk aversion in the gain domain in the immediate aftermath of two publicized events (the announcement of the quarantine of the city of Wuhan and the death of Dr Li Wenliang, a media figure). We start from the intuition that informed people should react differently to the Covid-19 crisis than less informed people. Specifically, the media conveys information about the development of the coronavirus crisis and should reinforce emotional responses to the situation through increased risk perception, fear, and anxiety.

To shed some light on the role of media exposure, we ask participants to report the sources of information they use to keep up to date regarding the Covid-19 evolution and the measures and restrictions implemented by the government (Figure C2 and Table C4). We first interact our Post-Covid dummy with the number of media used to be informed of the Covid-19 evolution. We report in Figure 5 only the interaction term. We find that the more a person uses multiple sources of information, the more strengthened is the impact of Covid-19 on preferences, both in the short and the medium term. Using one more information source increases risk aversion in the gain domain by an additional two percentage points. At the same time, it decreases it by a further four percentage points in the loss domain and increases impatience by an extra 1.6 percentage points. Besides, literature has highlighted the importance of information accuracy on individual beliefs and behaviors in the specific context of Covid-19.<sup>14</sup> Unfortunately, we are not able to capture the type of information received and how much attention they paid to it. Moreover, many unobservable behaviors can explain why people diversify their sources of information, such as the seeking out for correct or accurate conclusion or seeking confirmation to prior beliefs. In contrast, a low level of

<sup>&</sup>lt;sup>14</sup>Gutierrez et al. (2022) randomized information about the epidemic in an online survey in Mexico. They show that inaccurate real-time information due to reporting death delays leads to individuals being slower to adopt protective behaviors and alter their perceptions regarding the severity of the epidemic. Simonov et al. (2020) also find that a one percentage point increase in Fox News viewership, a leading news media channel that denied expert recommendations from the global health communities and minimized the danger of Coronavirus, reduced the propensity to stay at home by 8.9 percentage points. Bursztyn et al. (2020) examine how exposure to different informational content - even in a short-term period-drives beliefs, behavior, and downstream health outcomes and find differences in the timing of adoption of cautious behavior according to which shows is viewed.

information may reflect some distrust in the information disclosed or some backfiring effect (when respondents more strongly endorse a misperception about a controversial issue if their beliefs or predispositions are challenged). We do, however, have data on the sources of information (Figure 5). Social media, for instance, is considered to have a faster diffusion of information and increase fear and anger (Depoux et al., 2020). We first find that governmental and social media (Facebook and WhatApps) have strengthened effects on the instability of preferences due to the Covid-19. These results are in line with previous research on the impact of official information disclosure on behaviors (Banerjee et al., 2019) and on social media (Oh et al., 2021; Zeballos Rivas et al., 2021).



Figure 5: Heterogeneity according to the sources of media - Short-term sample

Author's estimations for the period January 2020 to June 2020. Only the interaction terms is represented in this figure.

## 6 Limitations

**External validity.** Our sample comprises women over 18 working in the informal sector in urban areas of Burkina Faso (Ouagadougou). The external validity of our results is questionable. The informal sector constitutes nearly 95% of the national labor force (BIT, 2019) and 55% of people in working age (18-65 years old) are self-employed.<sup>15</sup> Therefore, our parent population (women working as self-employed) represents 31% of the Burkinabe population.

One might question the extent to which a male population can be compared to a female population. Indeed, women were more affected by the Covid-19 crisis than men.<sup>16</sup> Alfonsi et al. (2022) shows that in the Ugandan urban context, high-skilled female workers suffered more than high-skilled male workers from the Covid-19 shutdown. Specifically, they show that lockdowns reduce women's employment more than men's and that after 18 months, male workers had returned to their prepandemic jobs. A proportion of female workers had permanently left their jobs or moved on to other opportunities less in line with their skills. Nevertheless, the specificity of their population makes it difficult to make any claims regarding ours. We checked in our overall sample whether women's activity decreased significantly more than men's, which is not the case. Thus, it is unlikely that their preferences are more malleable than men's concerning Covid-19.

Besides, risk aversion (without exposure to shocks) tends to be higher for women than men. Mathonnat et al. (2021) confirms this trend in May/June 2020 in Burkina Faso, using a self-reported measure. However, the literature is pretty silent on the potential malleability of gender preferences following a negative exogenous shock. Because women's initial level of risk aversion is higher than men's, the effect of a shock may be greater for men because they have more room to change. Some studies on the instability of risk aversion are disaggregated by gender but yield conflicting results: Hanaoka et al. (2018) suggests that men's risk aversion is more likely to change than women's after an earthquake in Japan, while Zhang and Palma (2022), studying the effect of the first Covid-19 lockdown in China, finds that the differential results vary depending on the method used to measure risk preferences. As a result, we cannot determine whether preference instability might have been greater for men had they been in our sample. In the same vein, the literature is not explicit about the level of entrepreneurial preference. From the entrepreneur's point of view, after obtaining the loan, borrowing agents are likely to make riskier decisions than they would otherwise have done. However, the argument is that the opposite can happen if the repayment mechanisms are so drastic,

 $<sup>^{15} {\</sup>rm Author's\ calculation\ from\ Enquête\ Harmonisée\ sur\ les\ Conditions\ de\ Vie\ des\ Ménages\ 2018-2019\ (EHCVM\ 18-19, https://microdata.worldbank.org/index.php/catalog/4290/related-materials$ 

<sup>&</sup>lt;sup>16</sup>Ahinkorah et al. (2021) overviews how the Covid-19 crisis has deepened gender inequalities in Sub-Saharan Africa.

making microentrepreneurs more risk-averse. As a conclusion, we cannot conclude on the changing preferences of non-recipients or non-entrepreneurs.

**Causal inference of the midterm results** As we briefly discussed in the empirical strategy section, our midterm results may not be, or may not be solely, the causal effect of the Covid-19 crisis in Burkina Faso. For several years, Burkina Faso has suffered from increasing insecurity over most of its territory. This long-lasting insecurity crisis has had dramatic consequences: millions of inhabitants have had to flee their homes to live in refugee camps or host households; inflation is very high and has caused a severe food crisis; political instability is due to the inability of governments to restore security. Each event is likely to redefine individuals' preferences: poor economic conditions affect background risk and risk perception, while exposure and testimony affect preferences. For example, Rockmore and Barrett (2022) use exposure to conflict in northern Uganda to show that receiving violence directly or indirectly-a family member or relative-increases risk-loving attitudes while witnessing or perpetuating violence increases negative attitudes toward risk. In such a context, it is difficult to predict whether our results underestimate or overestimate the true effect of Covid-19, but one can safely state that several mechanisms were at play simultaneously and are difficult to disentangle.

Yet, several elements tend to qualify the role of insecurity on preference change. First, the deterioration of the security situation in Burkina Faso has been more of an ongoing process since 2018 than a temporally situated event. Thus, this does not coincide with our results, namely a clear change between January and June 2020, with an effect maintained but not increased over time. Second, the political instability symbolized by the January 2022 *coup d'état* did not impact preference in the very short run. Finally, because our endline data collection was ongoing when the *coup* occurred in Ouagadougou, we can test, albeit only suggestively, whether respondents' preferences were sensitive to its occurrence. A simple discontinuity estimate of the regression (with the current variable being the date of the interviews) shows no discontinuity in the different types of preferences around the coup.

## 7 Conclusion and further research

In this study, we take advantage of the Covid-19 epidemic to empirically test whether risk and time preferences remain stable over time. The previous literature on the impact of shocks on preference stability focuses on macroeconomic shocks, natural disasters, or conflicts (Chuang and Schechter, 2015). The Covid-19 shock deserves an appropriate analysis because of its specific characteristics: it is a singular life-threatening shock, causing anxiety, with long-term consequences. Unlike regular economic downturns or frequent weather shocks, people may change their preferences more substantially as a result of an infrequent and striking event with unpredictable consequences. More importantly, the Covid-19 crisis was unexpected, uninsurable, and exogenous: it is a quasi-perfect natural experiment. We rely on a unique panel data set that allows us to track changes in the risk and time preferences of the same individual, observed just before the Covid-19 pandemic (January 2020), six months after (June 2020) and then two years later in Ouagadougou (Burkina Faso). We combine before-and-after comparisons with individual fixed effects to isolate the specific impact of Covid-19 on preference instability. This panel data allows us to provide a clear identification, with two different timespans.

This paper demonstrates that individual preferences have changed following the Covid-19 crisis, and that those changes are still observed two years later. Specifically, we show a 12% increase in risk aversion in the gain domain between January and June 2020 and a 47% decrease in risk aversion in the loss domain over the same period. Impatience increases by 20%. The results are robust to alternative specifications and outcome definitions. Eighteen months later, we observe that individual preferences have not returned to their initial level before the Covid-19 shock, which indicates that the impact of the pandemic on preferences was both immediate and long-lasting. We also try to identify the mechanisms underlying this preference instability. Using interactions, we show that preference variability is weakly influenced by the actual consequences experienced by respondents during the Covid-19 crisis, such as loss of a job, lack of food and water, or poor health in general, or symptoms of the Covid-19 crisis. In contrast, preference instability is exacerbated when respondents express strong concerns about the economic aspects of the Covid-19 crisis, such as fear of running out of money or economic collapse. Fears related to health aspects, such as spreading or contracting the disease, had no effect. These results suggest that the emotional channel has much more influence on preference instability than the actual consequences of the Covid-19 crisis. A second piece of evidence for the prominent role of emotion is highlighted by media exposure, which can increase anxiety and fear, and thus enhance the emotional response. The more informed a person is, the more unstable their preferences become. Similarly, we report that these effects are enhanced when she uses social networks (WhatsApp and Facebook) to keep up with health and economic developments in Covid-19. Our study includes limitations, mainly linked to the external validity of our sample, constituted of urban female informal workers, and to the existence of several other potential causes of preferences' shift in our two-year span analyses.

Increasing empirical evidence shows that preferences are not static and can change over time. The

stability of preferences is a fundamental principle of economic theory, but it is also practical from an empirical point of view. Indeed, assuming stability means that preferences are exogenous to any outcome of interest. Our results show that preferences change in response to life shocks. This implies that empirical inferences are biased with potential reverse causality and simultaneity bias ((Cobb-Clark and Schurer, 2012) provide a thorough discussion of endogeneity problems when the assumption of preference stability does not hold). In addition, preference instability has implications for policy design. If policies are based on the assumption of stable risk and time preferences, it may be challenging to predict appropriate behavioral responses in a post-shock period. For instance, the government or lenders may provide inadequate lending opportunities to entrepreneurs if they do not consider changes in economic agents' risk and time preferences. For example, our study shows that individuals exhibit greater impatience and a more risk-lover attitude toward losses. Thus, they may have been less inclined to use insurance since the onset of the Covid-19 crisis (). Given the low adoption of voluntary insurance in developing countries, our results could lead to advocating for insurance models that encourage people to enroll (such as mandatory enrollment). Finally, using the uniqueness of the Covid crisis, our study establishes causal evidence that preferences can change rapidly in response to a shock and remain permanently at this new level. This non-return to normal is all the more surprising given that we identify the emotional channel as the primary driver of these changes. Whereas previous literature suggests that emotional instability of preferences is temporary Schildberg-Hörisch (2018), our results suggest that the emotional response to the Covid-19 crisis altered, at least in the medium term, the perception of global and/or background risk (Ho et al., 2008; Lerner et al., 2015; Brown et al., 2018). Understanding the mechanisms underlying these changes is critical to designing effective policies and programs. Thus, future research should focus on better understanding the role of emotions and concurrently on the issue of identifying these emotional mechanisms in a household survey.

## Authors statement

## Declaration of competing interest

We, Delphine Boutin, Haris Megzari and Laurene Petitfour, the authors of the manuscript entitled "Instability of preferences due to Covid-19 Crisis and emotions: a natural experiment from urban Burkina Faso", gratefully acknowledge financial support from Bordeaux School of Economics (University of Bordeaux) and from Agence Française du Développement (AFD). We have no material or financial interest in this investigation or any other funding to disclose. The findings expressed in this paper are those of the authors and do not necessarily represent the views of the IMF or its partners.

## **Pre-registration**

The trial was pre-registered with the American Economic Association's Trial Registry https://www.socialscienceregistry.org/trials/8548.

## Ethics

This study was approved by the *Comité d'éthique institutionnel pour la recherche en Sciences de la santé- IRSS A027-2019.* All study participants provided informed consent.

## Contributorship statement

Delphine Boutin: Design the study; Data collection; Funding acquisition; Methodology; Literature review, Data cleaning, Data analysis, Formal analysis; Writing - original draft. Laurène Petifour: Design the study; Data collection; Funding acquisition; Data cleaning, Data analysis Haris Megzari: Literature review, Data cleaning Delphine Boutin and Laurène Petitfour designed the study, sought funding and conducted data collection. Delphine Boutin, Laurène Petitfour and Haris Megzari performed the data analysis. Delphine Boutin wrote the manuscript. Delphine Boutin, Laurène Petitfour and Haris Megzari agreed with the submitted manuscript.

## Data availability

Anonymized data will be made available upon request.

## References

- Adjei-Mantey, K. and Horioka, C. Y. Determinants of health insurance enrollment and health expenditure in ghana: An empirical analysis. *Review of Economics of the Household*.
- Ahinkorah, B. O., Hagan, J. E., Ameyaw, E. K., Seidu, A.-A., and Schack, T. (2021). Covid-19 pandemic worsening gender inequalities for women and girls in sub-saharan africa. Frontiers in Global Women's Health, 2.
- Alfonsi, L., Namubiru, M., and Spaziani, S. (2022). Gender gaps: Back and here to stay? evidence from skilled ugandan workers during covid-19.
- Andersen, S., Harrison, G. W., Lau, M. I., and Rutström, E. E. (2008). Eliciting risk and time preferences. *Econometrica*, 76(3):583–618.
- Andersson, O., Holm, H. J., Tyran, J.-R., and Wengström, E. (2020). Robust inference in risk elicitation tasks. *Journal of Risk and Uncertainty*, pages 1–15.
- Angrisani, M., Cipriani, M., Guarino, A., Kendall, R., and Ortiz de Zarate, J. (2020). Risk preferences at the time of covid-19: An experiment with professional traders and students. FRB of New York Staff Report, (927).
- Aragon, F., Bernal, N., Bosch, M., Molina, O., et al. (2022). Covid-19 and economic preferences: evidence from a panel of cab drivers. Technical report.
- Banerjee, A., Alsan, M., Breza, E., Chandrasekhar, A. G., Chowdhury, A., Duflo, E., Goldsmith-Pinkham, P., and Olken, B. A. (2020). Messages on covid-19 prevention in india increased symptoms reporting and adherence to preventive behaviors among 25 million recipients with similar effects on non-recipient members of their communities. Technical report, National Bureau of Economic Research.
- Banerjee, A., Ferrara, E. L., and Orozco-Olvera, V. H. (2019). The entertaining way to behavioral change: Fighting hiv with mtv. Technical report, National Bureau of Economic Research.
- Barberis, N., Huang, M., and Santos, T. (2001). Prospect theory and asset prices. The quarterly journal of economics, 116(1):1–53.
- Bargain, O. and Aminjonov, U. (2021). Poverty and covid-19 in africa and latin america. World development, 142:105422.

- Barsky, R. B., Juster, F. T., Kimball, M. S., and Shapiro, M. D. (1997). Preference parameters and behavioral heterogeneity: An experimental approach in the health and retirement study. *The Quarterly Journal of Economics*, 112(2):537–579.
- Baucells, M. and Villasís, A. (2010). Stability of risk preferences and the reflection effect of prospect theory. *Theory and Decision*, 68(1-2):193–211.
- Bchir, M. A., Willinger, M., et al. (2013). Does the exposure to natural hazards affect risk and time preferences? some insights from a field experiment in perú. *Unpublished Manuscript*.
- Biener, C. and Landmann, A. (2023). Recovery mode: Non-cognitive skills after the storm. World Development, 164:106152.
- BIT (2019). Femmes et hommes dans l'économie informelle: un panorama statistique.
- Bordalo, P., Gennaioli, N., and Shleifer, A. (2012). Salience theory of choice under risk. *The Quarterly journal of economics*, 127(3):1243–1285.
- Botzen, W. W., Kunreuther, H., and Michel-Kerjan, E. (2015). Divergence between individual perceptions and objective indicators of tail risks: Evidence from floodplain residents in new york city. Judgment and Decision Making, 10(4):365–385.
- Brown, P., Daigneault, A. J., Tjernström, E., and Zou, W. (2018). Natural disasters, social protection, and risk perceptions. World development, 104:310–325.
- Bu, D., Hanspal, T., Liao, Y., and Liu, Y. (2020). Risk taking during a global crisis: Evidence from wuhan. *Covid Economics*, 5:106–146.
- Bursztyn, L., Rao, A., Roth, C. P., and Yanagizawa-Drott, D. H. (2020). Misinformation during a pandemic. Technical report, National Bureau of Economic Research.
- Callen, M. (2015). Catastrophes and time preference: Evidence from the indian ocean earthquake. Journal of Economic Behavior & Organization, 118:199–214.
- Cameron, L. and Shah, M. (2015). Risk-taking behavior in the wake of natural disasters. Journal of Human Resources, 50(2):484–515.
- Campbell, J. Y. and Cochrane, J. H. (1999). By force of habit: A consumption-based explanation of aggregate stock market behavior. *Journal of political Economy*, 107(2):205–251.
- Cassar, A., Healy, A., and Von Kessler, C. (2017). Trust, risk, and time preferences after a natural disaster: experimental evidence from thailand. *World Development*, 94:90–105.

- Castillo, J. G. and Hernandez, M. A. (2022). The unintended consequences of confinement: Evidence from the rural area in guatemala. *Journal of Economic Psychology*, page 102587.
- Choi, S., Kariv, S., Müller, W., and Silverman, D. (2014). Who is (more) rational? *American Economic Review*, 104(6):1518–50.
- Chuang, Y. and Schechter, L. (2015). Stability of experimental and survey measures of risk, time, and social preferences: A review and some new results. *Journal of development economics*, 117:151–170.
- Cobb-Clark, D. A. and Schurer, S. (2012). The stability of big-five personality traits. *Economics* Letters, 115(1):11–15.
- Cohn, A., Engelmann, J., Fehr, E., and Maréchal, M. A. (2015). Evidence for countercyclical risk aversion: An experiment with financial professionals. *American Economic Review*, 105(2):860–85.
- Crepaz, M. and Arikan, G. (2021). Information disclosure and political trust during the covid-19 crisis: experimental evidence from ireland. *Journal of Elections, Public Opinion and Parties*, 31(sup1):96–108.
- Dalton, P. S., Nhung, N., and Rüschenpöhler, J. (2020). Worries of the poor: The impact of financial burden on the risk attitudes of micro-entrepreneurs. *Journal of Economic Psychology*, 79:102198.
- Dawson, C. and Henley, A. (2015). Gender, risk, and venture creation intentions. Journal of Small Business Management, 53(2):501–515.
- Depoux, A., Martin, S., Karafillakis, E., Preet, R., Wilder-Smith, A., and Larson, H. (2020). The pandemic of social media panic travels faster than the covid-19 outbreak. *Journal of travel medicine*, 27(3):taaa031.
- Dohmen, T., Falk, A., Golsteyn, B. H., Huffman, D., and Sunde, U. (2017). Risk attitudes across the life course. *Economic Journal*, 127(605):F95–F116.
- Dohmen, T., Falk, A., Huffman, D., and Sunde, U. (2010). Are risk aversion and impatience related to cognitive ability? *American Economic Review*, 100(3):1238–60.
- Dohmen, T., Falk, A., Huffman, D., Sunde, U., Schupp, J., and Wagner, G. G. (2011). Individual risk attitudes: Measurement, determinants, and behavioral consequences. *Journal of the European Economic Association*, 9(3):522–550.

- Drichoutis, A. C. and Nayga, R. M. (2022). On the stability of risk and time preferences amid the covid-19 pandemic. *Experimental Economics*, 25(3):759–794.
- Dupas, P. (2011). Health behavior in developing countries. Annu. Rev. Econ., 3(1):425–449.
- Eckel, C. C., El-Gamal, M. A., and Wilson, R. K. (2009). Risk loving after the storm: A bayesiannetwork study of hurricane katrina evacuees. *Journal of Economic Behavior & Organization*, 69(2):110–124.
- Eeckhoudt, L., Gollier, C., and Schlesinger, H. (1996). Changes in background risk and risk taking behavior. *Econometrica: Journal of the Econometric Society*, pages 683–689.
- Fang, G., Li, W., and Zhu, Y. (2022). The shadow of the epidemic: Long-term impacts of meningitis exposure on risk preference and behaviors. *World Development*, 157:105937.
- Fetzer, T., Hensel, L., Hermle, J., and Roth, C. (2021). Coronavirus perceptions and economic anxiety. *Review of Economics and Statistics*, 103(5):968–978.
- Fleeson, W. (2001). Toward a structure-and process-integrated view of personality: Traits as density distributions of states. *Journal of personality and social psychology*, 80(6):1011.
- Galandra, C., Cerami, C., Santi, G. C., Dodich, A., Cappa, S. F., Vecchi, T., and Crespi, C. (2020). Job loss and health threatening events modulate risk-taking behaviours in the covid-19 emergency. *Scientific reports*, 10(1):1–10.
- Gassmann, X., Malézieux, A., Spiegelman, E., and Tisserand, J.-C. (2022). Preferences after pan (dem) ics: Time and risk in the shadow of covid-19. Technical Report 4.
- Gollier, C. and Pratt, J. W. (1996). Risk vulnerability and the tempering effect of background risk. Econometrica: Journal of the Econometric Society, pages 1109–1123.
- Gonzalez, C., Dana, J., Koshino, H., and Just, M. (2005). The framing effect and risky decisions: Examining cognitive functions with fmri. *Journal of economic psychology*, 26(1):1–20.
- Guiso, L. and Paiella, M. (2008). Risk aversion, wealth, and background risk. Journal of the European Economic association, 6(6):1109–1150.
- Guiso, L., Sapienza, P., and Zingales, L. (2018). Time varying risk aversion. Journal of Financial Economics, 128(3):403–421.

- Gutierrez, E., Rubli, A., and Tavares, T. (2022). Information and behavioral responses during a pandemic: evidence from delays in covid-19 death reports. *Journal of Development Economics*, 154:102774.
- Hanaoka, C., Shigeoka, H., and Watanabe, Y. (2018). Do risk preferences change? evidence from the great east japan earthquake. American Economic Journal: Applied Economics, 10(2):298– 330.
- Harrison, G. W., Hofmeyr, A., Kincaid, H., Monroe, B., Ross, D., Schneider, M., and Swarthout, J. T. (2022). Subjective beliefs and economic preferences during the covid-19 pandemic. *Experimental economics*, pages 1–29.
- Hetschko, C. and Preuss, M. (2020). Income in jeopardy: How losing employment affects the willingness to take risks. *Journal of Economic Psychology*, 79:102175.
- Ho, M.-C., Shaw, D., Lin, S., and Chiu, Y.-C. (2008). How do disaster characteristics influence risk perception? *Risk Analysis: An International Journal*, 28(3):635–643.
- Holt, C. A. and Laury, S. K. (2002). Risk aversion and incentive effects. American Economic Review, 92(5):1644–1655.
- Hsieh, C., Parker, S. C., and van Praag, C. M. (2017). Risk, balanced skills and entrepreneurship. Small Business Economics, 48(2):287–302.
- Ikeda, S., Yamamura, E., Tsutsui, Y., et al. (2020). Covid-19 enhanced diminishing sensitivity in prospect-theory risk preferences: A panel analysis. *Institute of Social and Economic Research Discussion Papers*, 1106:1–25.
- Kahneman, D. and Tversky, A. (1979). Prospect theory: An analysis of decision under risk. *Econometrica*, 47 (2):263–291.
- Kahneman, D. and Tversky, A. (2013). Prospect theory: An analysis of decision under risk. In Handbook of the fundamentals of financial decision making: Part I, pages 99–127. World Scientific.
- Kahsay, G. A. and Osberghaus, D. (2016). Extreme weather and risk preference: Panel evidence from germany. ZEW Discussion Papers, 16.
- Kimball, M. S. (1993). Standard risk aversion. Econometrica: Journal of the Econometric Society, pages 589–611.

- Kobiane, J.-F. (2020). Les inégalités au burkina faso à l'aune de la pandémie de la covid-19: quelques réflexions prospectives. Technical report, Agence française de développement.
- Kremer, M., Rao, G., and Schilbach, F. (2019). Behavioral development economics. In Handbook of Behavioral Economics: Applications and Foundations 1, volume 2, pages 345–458. Elsevier.
- La Ferrara, E. (2016). Mass media and social change: Can we use television to fight poverty? Journal of the European Economic Association, 14(4):791–827.
- La Ferrara, E., Chong, A., and Duryea, S. (2012). Soap operas and fertility: Evidence from brazil. American Economic Journal: Applied Economics, 4(4):1–31.
- Laury, S. K. and Holt, C. A. (2008). Further reflections on the reflection effect. In *Risk aversion in experiments*. Emerald Group Publishing Limited.
- Lerner, J. S. and Keltner, D. (2001). Fear, anger, and risk. Journal of personality and social psychology, 81(1):146.
- Lerner, J. S., Li, Y., Valdesolo, P., and Kassam, K. S. (2015). Emotion and decision making. Annual review of psychology, 66.
- Levin, I. P., Xue, G., Weller, J. A., Reimann, M., Lauriola, M., and Bechara, A. (2012). A neuropsychological approach to understanding risk-taking for potential gains and losses. *Frontiers* in neuroscience, 6:15.
- Li, J.-Z., Li, S., Wang, W.-Z., Rao, L.-L., and Liu, H. (2011). Are people always more risk averse after disasters? surveys after a heavy snow-hit and a major earthquake in china in 2008. *Applied cognitive psychology*, 25(1):104–111.
- Li, K. K., Huang, B., Tam, T., and Hong, Y.-y. (2020). Does the covid-19 pandemic affect people's social and economic preferences? evidence from china. *Evidence from China (July 6, 2020)*.
- Loewenstein, G. (2000). Emotions in economic theory and economic behavior. *American economic review*, 90(2):426–432.
- Loewenstein, G. and Lerner, J. S. (2003). The role of affect in decision making.
- Lopes, L. L. (1987). Between hope and fear: The psychology of risk. Advances in experimental social psychology, 20(3):255–295.

- Mahmud, M. and Riley, E. (2020). Household response to an extreme shock: Evidence on the immediate impact of the covid-19 lockdown on economic outcomes and well-being in rural uganda. World Development, 140:105318.
- Mathonnat, J., Audibert, M., and Nossek, V. (2021). Perception de la covid et comportement des ménages au burkina faso. une analyse sur données d'enquête.
- Mosley, P. and Verschoor, A. (2005). Risk attitudes and the 'vicious circle of poverty'. *The European* journal of development research, 17(1):59–88.
- Oh, S.-H., Lee, S. Y., and Han, C. (2021). The effects of social media use on preventive behaviors during infectious disease outbreaks: The mediating role of self-relevant emotions and public risk perception. *Health communication*, 36(8):972–981.
- Pachur, T., Schulte-Mecklenbeck, M., Murphy, R. O., and Hertwig, R. (2018). Prospect theory reflects selective allocation of attention. *Journal of Experimental Psychology: General*, 147(2):147.
- Price, R. (2020). Impacts of covid-19 regulatory measures on small-scale and informal trade in zimbabwe.
- Reynaud, A. and Aubert, C. (2020). Does flood experience modify risk preferences? evidence from an artefactual field experiment in vietnam. *The Geneva Risk and Insurance Review*, 45(1):36–74.
- Rockmore, M. and Barrett, C. B. (2022). The implications of aggregate measures of exposure to violence for the estimated impacts on individual risk preferences. *World Development*, 157:105925.
- Sadish, D., Adhvaryu, A., and Nyshadham, A. (2021). (mis) information and anxiety: Evidence from a randomized covid-19 information campaign. *Journal of Development Economics*, page 102699.
- Sakha, S. (2019). Determinants of risk aversion over time: Experimental evidence from rural thailand. *Journal of Behavioral and Experimental Economics*, 80:184–198.
- Schildberg-Hörisch, H. (2018). Are risk preferences stable? Journal of Economic Perspectives, 32(2):135–54.
- Shachat, J., Walker, M. J., and Wei, L. (2021). The impact of an epidemic: Experimental evidence on preference stability from wuhan. AEA Papers and Proceedings, 111:302–06.
- Simonov, A., Sacher, S. K., Dubé, J.-P. H., and Biswas, S. (2020). The persuasive effect of fox news: non-compliance with social distancing during the covid-19 pandemic. Technical report, National Bureau of Economic Research.

- Stigler, G. J. and Becker, G. S. (1977). De gustibus non est disputandum. The american economic review, 67(2):76–90.
- Tice, D. M., Bratslavsky, E., and Baumeister, R. F. (2001). Emotional distress regulation takes precedence over impulse control: If you feel bad, do it! *Journal of personality and social psychology*, 80(1):53.
- Tymula, A., Belmaker, L. A. R., Ruderman, L., Glimcher, P. W., and Levy, I. (2013). Like cognitive function, decision making across the life span shows profound age-related changes. *Proceedings* of the National Academy of Sciences, 110(42):17143–17148.
- Van Der Pol, M., Hennessy, D., and Manns, B. (2017). The role of time and risk preferences in adherence to physician advice on health behavior change. *The European Journal of Health Economics*, 18(3):373–386.
- Voors, M. J., Nillesen, E. E., Verwimp, P., Bulte, E. H., Lensink, R., and Van Soest, D. P. (2012). Violent conflict and behavior: a field experiment in burundi. *American Economic Review*, 102(2):941–64.
- Willemsen, M. C., Böckenholt, U., and Johnson, E. J. (2011). Choice by value encoding and value construction: Processes of loss aversion. *Journal of Experimental Psychology: General*, 140(3):303.
- Yesuf, M. and Bluffstone, R. A. (2009). Poverty, risk aversion, and path dependence in low-income countries: Experimental evidence from ethiopia. *American Journal of Agricultural Economics*, 91(4):1022–1037.
- Zarocostas, J. (2020). How to fight an infodemic. The lancet, 395(10225):676.
- Zeballos Rivas, D. R., Lopez Jaldin, M. L., Nina Canaviri, B., Portugal Escalante, L. F., Alanes Fernández, A. M., and Aguilar Ticona, J. P. (2021). Social media exposure, risk perception, preventive behaviors and attitudes during the covid-19 epidemic in la paz, bolivia: A cross sectional study. *PloS one*, 16(1):e0245859.
- Zhang, P. and Palma, M. A. (2022). Stability of risk preferences during covid-19: Evidence from four measurements. *Frontiers in psychology*, 12:6619.

# Appendix A: Risk and time preference measurement

Figure A1: Elicitation of risk attitudes in the gain domain

# Imagine you have to choose between these two options, A or B. Which option would you choose ?



Table A1: Measurement of risk attitudes in the gain domain

	Scenario 1	Scenario 2	Scenario 3
0- High risk taking	В	В	В
1	В	В	А
2	В	А	А
3- High risk aversion	А	А	А

Figure A2: Elicitation of risk attitudes in the loss domain

Imagine you have to choose between these two options, A or B. Which option would you choose ?



Table A2: Measurement of risk attitudes in the loss domain

	Scenario 1	Scenario 2	Scenario 3
0- High risk taking	В	В	В
1	В	В	А
2	В	А	А
3- High risk aversion	А	А	А

Figure A3: Elicitation of impatience

# Imagine you have to choose between these two options, A or B. Which option would you choose ?

Scenario 1	Option A		Option B
	5000 FCFA now		20000 FCFA in 1 month
Scenario 2	Option A	or	Option B
	5000 FCFA now		20000 FCFA in 3 months
Scenario 3	Option A	or	Option B
	5000 FCFA now		20000 FCFA in 6 months

Table A3: Measurement of impatience

	Scenario 1	Scenario 2	Scenario 3
0- High patience	В	В	В
1	В	В	А
2	В	А	А
3- High impatience	А	А	А

	Balanced panel on two waves		Balanced panel on three waves		
	Jan 2020	June 2020	Jan 2020	June 2020	Jan 2022
0 - High risk-taking	23.01	14.64	25.34	12.57	17.21
1 - Moderate risk taking	2.10	0.94	2.48	0.56	3.55
2 - Moderate risk aversion	0.35	0.94	0.28	1.4	5.19
3 - High risk aversion	74.53	83.49	71.90	85.47	<b>74.04</b>

Table A4: Descriptive statistics for risk aversion (gain domain)

Table A5: Descriptive statistics for risk aversion (loss domain)

	Balanced panel on two waves		Balanced panel on three waves			
	Jan 2020	June 2020	Jan 2020	June 2020	Jan 2022	
0 - High risk-taking	36.69	68.11	35.77	69.97	48.49	
1 - Moderate risk taking	0.47	1.05	0.28	1.38	4.38	
2 - Moderate risk aversion	2.72	0.47	2.82	0.28	10.41	
<b>3</b> - High risk aversion	60.12	30.37	61.13	28.37	36.71	

Table A6: Descriptive statistics for present preference

	Balanced p	anel on two waves	Balanced panel on three waves			
	Jan 2020 June 2020		Jan 2020	June 2020	Jan 2022	
0 - High patience	14.15	9.05	11.33	7.58	8.72	
1 - Moderate patience	8.47	4.58	8.84	4.78	4.36	
2 - Moderate impatience	12.3	6.93	12.15	4.49	10.63	
3 - High impatience	65.08	79.44	67.68	83.15	76.29	

Figure A4: Example of visual support during the survey to facilitate the choice



Figure A5: Example of visual support of comprehension question



## Appendix B: Test for selective difference in samples

	Baseline	Midline	Endline
Period	January - February	May- June	January - February
Year	2020	2020	2022
Survey type	Face-to-face	Phone	Face-to-face
Initial sample size	1832	871	1559
Short-term sample size	871	871	-
Medium-term sample size	366	366	366

Table B1: Samples characteristics

Table B1 explains the different types of surveys. Initially, we had 1832 women interviewed in the baseline survey (randomly selected from approximately 10,000 women microentrepreneurs based in Ouagadougou). As explained in the data section, we could only interview half of these 1,839 women (randomly selected for the interim survey) in the midline survey due to budgetary constraints. Thus, the 871 women interviewed in the midline survey are balanced with our baseline survey. Our short-term sample thus includes these 871 women, interviewed twice (baseline and intermediate survey).

In the final survey, we tried to find as many of the 1,832 women interviewed in the baseline survey as possible. We were able to locate 838 of these women, 366 of whom had also been interviewed in the midline survey. The microcredit offered to these women is solidarity-based, in the sense that they must group, with group sizes ranging from 10 to 40 people. Therefore, we replaced the 1,228 attritors with 903 women who belonged to the exact same microcredit group from which the attritors came. Our total sample size for the final survey is, therefore, 1559 women. For this study, however, our medium-term sample includes only those women who were actually observed three times.

	Jan 2	2020 only	Jan 202	20 and June 2022	Differ	ence
	N	=961		N=871		
	Mean	St. Dev.	Mean	St. Dev.	Raw diff.	P-value
Time-invariant characteristics						
Married	87.41	0.33	86.34	0.34	1.07	0.498
Ethnics: Mossi	87.1	0.34	83.35	0.37	3.74	0.024
Muslim	66.81	0.47	69.23	0.46	-2.43	0.267
Catholic	25.81	0.44	24.11	0.43	1.7	0.403
Not Burkinabe	10.09	0.30	10.68	0.31	-0.58	0.683
Cannot read or write	32.47	0.47	33.98	0.47	-1.52	0.491
Never been in school	63.68	0.48	60.28	0.49	3.41	0.133
Primary level of education	24.56	0.43	27.67	0.45	-3.11	0.130
Covid-19 vulnerable characteristics						
Has a job (last week)	91.13	0.28	92.36	0.27	-1.23	0.343
Has a formal job	3.02	0.17	4.59	0.21	-1.57	0.077
Sector of activity: small business	86.99	0.34	85.65	0.35	1.34	0.403
Has a local or specific place for her activity	44.95	0.50	41.68	0.49	3.28	0.158
Has not been sick (last 3 months)	50.99	0.50	51.89	0.50	-0.91	0.699
Use of healthcare if sick	87.82	0.33	89.86	0.30	-2.04	0.333
Financial situation						
Has a credit	80.75	0.39	81.63	0.39	-0.88	0.630
Has a credit for less than 2 years	74.19	0.44	74.37	0.44	-0.17	0.932
Use credit for business	80.02	0.40	81.4	0.39	-1.38	0.455
Has a moto	87.72	0.33	85.19	0.36	2.53	0.113
Own some cultivable land	8.84	0.28	7.12	0.26	1.73	0.174
Has a tv	50.16	0.50	54.42	0.50	-4.26	0.068
Has a phone	98.54	0.12	99.66	0.06	-1.11	0.013
Kisk and time preferences	70.11	0.44	74 50	0.44	1.40	0.405
Kisk averse (gain domain)	73.11	0.44	74.53	0.44	-1.42	0.495
Kisk averse (loss domain)	60.85	0.49	60.12	0.49	0.73	0.752
High impatient	67.74	0.47	64.41	0.48	3.33	0.132

## Table B2: Test for selective difference in two-waves sample (baseline and midline)

Joint F-test for orthogonality : 0.19

Note: In this two-wave sample, there is no attrition. We were not able to re-interview all respondents in the database due to financial constraints. We randomly selected nearly half of the participants and were able to interview all those selected.

	Jan 2020 only		Jan 2020 and June 2022		Differ	ence
	N	=961		N=871	D 1.0	
<i>—</i>	Mean	St. Dev.	Mean	St. Dev.	Raw diff.	P-value
Time-invariant characteristics	~~ ~ .	0.0 <b>×</b>	0= 10	0.04	1.10	0 5 40
Married	85.74	0.35	87.16	0.34	-1.42	0.549
Ethnics: Mossi	83.96	0.37	82.51	0.38	1.45	0.572
Muslim	71.09	0.45	66.67	0.47	4.42	0.163
Catholic	21.78	0.41	27.32	0.45	-5.54	0.059
Not Burkinabe	12.28	0.33	8.47	0.28	3.81	0.073
Cannot read or write	33.66	0.47	34.43	0.48	-0.76	0.815
Never been in school	62.38	0.48	57.38	0.50	5	0.137
Primary level of education	24.75	0.43	31.69	0.47	-6.94	0.024
Covid-19 vulnerable characteristics						
Has a job (last week)	91.45	0.28	93.63	0.24	-2.18	0.235
Has a formal job	4.16	0.20	5.19	0.22	-1.03	0.473
Sector of activity: small business	87.33	0.33	83.33	0.37	3.99	0.097
Has a local or specific place for her activity	39.21	0.49	45.08	0.50	-5.87	0.083
Has not been sick (last 3 months)	47.72	0.50	57.65	0.49	-9.93	0.004
Use of healthcare if sick	90.53	0.29	88.75	0.32	1.78	0.557
Financial situation						
Has a credit	83.17	0.37	79.51	0.40	3.66	0.169
Has a credit for less than 2 years	75.2	0.43	73.22	0.44	1.97	0.511
Use credit for business	82.57	0.38	79.78	0.40	2.79	0.296
Has a moto	84.75	0.36	85.79	0.35	-1.04	0.670
Own some cultivable land	6.14	0.24	8.47	0.28	-2.33	0.187
Has a tv	53.66	0.50	55.46	0.50	-1.80	0.599
Has a phone	99.41	0.08	100	0.00	-0.59	0.140
-						
Risk and time preferences						
Risk averse (gain domain)	76.47	0.42	71.9	0.45	4.57	0.130
Risk averse (loss domain)	59.39	0.49	61.13	0.49	-1.74	0.611
High impatient	62.57	0.48	66.94	0.47	-4.37	0.185

Table B3: Test for selective attrition in three-waves sample (baseline, midline and endline)

Joint F-test for orthogonality : 0.00

Note: Note: In this three-wave sample, the attrition rate is 57.98%: among the 871 respondents surveyed in January and June 2020, we were able to reach only 42.02%.

## Appendix C: Coronavirus Crisis in Ouagadougou (Burkina Faso)

#### Suggestive evidence of compliance

The figure below shows the compliance of Burkinabe located in the Ouagadougou region with the measures put in place to contain COVID-19. Data comes from the Google Community Mobility dataset.<sup>17</sup>

Figure C1: Flow of people's movement since the first Covid-19 case in Burkina Faso



Authors from Google Community Mobility data in Ouagadougou region (Burkina Faso)

<sup>&</sup>lt;sup>17</sup>Using user's history data from mobile device location applications (such as Google Maps), Google aggregates anonymized datasets and publicly provides population movements throughout the pandemic. The data shows how visits to places, such as grocery stores and parks, are changing in each geographic region. Each day, the number of visitors to specific categories of places (e.g., grocery stores, parks, train stations) is measured and compared to a baseline day (estimated before the start of the pandemic over the five weeks from January 3 to February 6, 2020).

The remaining part of this Appendix B presents some descriptive statistics regarding the sources of information used by the respondents, their perceptions of the coronavirus crisis, and its impact on daily life.

#### Sources and types of information

Figure C2 shows the main sources of information used by respondents to keep up to date with Covid-19 and the measures and restrictions put in place by the government. Almost all respondents (91%) were getting information from listening to the radio and most (72%) from watching television. Official government messages (17%) and public signage (16%) were relatively unused communication channels. On the other hand, 33% of respondents got their information from meetings organized by MFIs. Physical and online social networks played an important role: 57% of respondents obtained information by talking with friends and family, and 12% and 13% through WhatsApp and Facebook.





More generally, 96% used an official source of information (government message, newspapers, radio, television, public signage), 66% exchanged information through discussions with family and friends

or at MFI meetings, and 14% through online social networks (Table C4). Most respondents use multiple sources of information. Only 11% use only one source, while 28% use two sources of information, 37% use three, and 24% use at least four. They also diversify the type of media they use, with only one-third of respondents getting their information exclusively from official channels and 4% only from discussions. It should be noted that over-information or diversification of information sources does not necessarily mean that the respondent will change their behavior. Indeed, some official messages were contradictory (e.g., those about wearing masks), and several false rumors (about the very existence of the virus or the immunity of blacks) circulated, which may have led to misinformation among respondents.

	Obs.	Freq.
Information sources		
$Official^1$	821	96.14
Discussions <sup>2</sup>	559	65.46
Online social networks <sup>3</sup>	117	13.70
$Official^1$ only	275	31.01
Discussions <sup>2</sup> only	30	3.51
Online social networks <sup>3</sup> only	0	0.00
Number of sources of information used		
1	94	10.79
2	240	28.10
3	313	36.65
4+	201	23.53

Table C4: Main sources of information about the coronavirus

Note: <sup>1</sup>Government messages, newspapers, radio, television, public signage. <sup>2</sup>Discussions within family, friends or at MFI's meetings. <sup>3</sup>WhatsApp and Facebook.

#### Concerns about Covid-19

The main concerns about the coronavirus crisis are primarily health-related (Table C5): 78% are afraid of contracting the disease and 58% are afraid of a family member contracting it. These concerns are considered the most worrisome (35% and 20% respectively). In addition, half of those surveyed (47%) said they fear the spread of the disease. 16% of people consider this to be the most

important concern. To a lesser extent, respondents also express concerns related to daily economic life: they fear not having enough money for food (36%), health care (29%) or credit (27%). Finally, a sizable proportion of respondents have more general concerns, such as fearing that they will no longer be able to travel (24%) or that the economy will collapse (21%). While most respondents express only one or two concerns (12% and 33%, respectively), a significant proportion (10%) say they are concerned about all the concerns mentioned.

	One concern		The most worrying concern	
	Obs.	Freq.	Obs.	Freq.
Contracting the disease	669	78.34	302	35.52
A family member contract the disease	499	58.43	175	20.11
The disease spread everywhere	403	47.19	138	16.18
Lack of money for food	305	35.71	72	8.44
Lack of money for health care	248	29.04	42	4.92
Lack of money for repay loans	230	26.93	58	6.67
Travel restrictions	201	23.54	32	3.75
The economy's collapse	175	20.49	28	3.28
Other	9	0.93	1	0.12

Table C5: Main concerns regarding the coronavirus crisis

#### The impacts of Covid-19 on daily life

The coronavirus crisis in Burkina Faso has impacted respondents' daily economic conditions. 35% of respondents regularly faced difficulties in obtaining food (several times during the period or systematically). In addition, 6% mentioned frequent shortages of drinking water and 5% cited shortages of medicine or difficulties accessing medical care (Table C6). In addition, 54% of respondents reported frequent restrictions in their economic activities and 7% reported systematic restrictions. When we asked respondents to self-assess their current living conditions in comparison to those of other Burkinabe (Table C7), the majority of individuals consider themselves to be in a better (52%) or much better (12%) situation than the rest of the population. However, more than a quarter of the respondents think their living conditions are worse (27%) or much worse (5%) than the rest of the population. Besides, 42% and 21% of respondents anticipated the economic conditions to improve in the year following the survey.

	Insufficient food	Lack of drinking water	Lack of medication or medical care	Restrictions on economic activities
Never	45.12	79.15	84.64	20.47
Just once or twice	20.45	14.25	10.81	18.12
Several times	32.90	4.27	2.99	54.63
Always	1.53	2.33	1.56	6.78
At least once	54.88	20.85	14.36	79.53
Obs.	851	772	768	767

Table C6: Impacts of Covid-19 on daily life

Table C7: Self-assessment of respondents' living conditions

How would you describe your current living conditions compared to those of other Burkinabes?

	Obs.	Freq.
Much Worse	45	5.28
Slightly worse	233	27.35
Same	24	2.82
Slightly better	444	52.11
Much better	106	12.44

## Appendix D. Robustness checks

	Short-term variation			Medium-term variation			
	(1)	(2)	(3)	(4)	(5)	(6)	
Risk preference (gain domain)							
Post	$0.269^{***}$	$0.274^{***}$	$0.284^{***}$	0.291***	0.292***	$0.276^{***}$	
	(0.055)	(0.056)	(0.058)	(0.079)	(0.079)	(0.083)	
Relative effect $(\%)$	11.88	11.93	12.37	13.29	13.26	12.80	
Nb. Observations	1710	1710	1604	1087	1087	1043	
Risk preference (loss domain)							
Post	-0.931***	-0.929***	-0.899***	-0.781***	$-0.784^{***}$	-0.803***	
	(0.066)	(0.067)	(0.075)	(0.086)	(0.087)	(0.092)	
Relative effect $(\%)$	-50.01	-49.93	-48.32	-41.23	-41.20	-42.26	
Nb. Observations	1701	1701	1596	1083	1083	1039	
Time preference							
Post	$0.284^{***}$	$0.279^{***}$	$0.245^{***}$	$0.226^{***}$	$0.226^{***}$	$0.197^{**}$	
	(0.045)	(0.045)	(0.051)	(0.059)	(0.059)	(0.064)	
Relative effect $(\%)$	12.42	12.26	10.51	9.57	9.74	8.47	
Nb. Observations	1713	1713	1585	1085	1085	1032	
Robust standard errors	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
Individual fixed-effect		$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$	
Control			$\checkmark$			$\checkmark$	
Nb. Observations	1710	1710	1604	1087	1087	1044	

Table D1: Tempora	l variation	of preferences	(Categorical	variable)
-------------------	-------------	----------------	--------------	-----------

Notes: Short-term variation : on balanced panel between baseline and midline.

Medium-term variation: on balanced panel between baseline, midline and endline.

Mean risk preference (gain domain) in baseline = 2.264 for the short-term sample and 2.1873 for the medium-term sample. Mean risk preference (loss domain) in baseline = 1.8627 for the short-term sample and 1.893 for the medium-term sample. Mean time preference in baseline = 2.2831 for the short-term sample and 2.3619 for the medium-term sample.

Post is a dummy variable indicating the second wave and the third wave.

Controls include having a job the week before the survey, age, level of education, asset score, household size, subjective health status.

Robust standard errors in parentheses. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

	Shor	rt-term varia	ation	Mediu	um-term var	iation
	(1)	(2)	(3)	(4)	(5)	(6)
Risk preference (gain domain)						
Post	0.092***	0.088***	0.100***	0.069**	0.055**	0.091***
	(0.023)	(0.019)	(0.021)	(0.028)	(0.028)	(0.030)
Relative effect $(\%)$	12.08	12.08	13.42	9.74	6.95	12.52
Nb. Observations	1604	1600	1517	1043	1043	994
Risk preference (loss domain)						
Post	-0.281***	-0.293***	-0.270***	-0.288***	-0.283***	-0.262***
	(0.023)	(0.024)	(0.027)	(0.031)	(0.030)	(0.032)
Relative effect $(\%)$	-46.57	-48.24	-44.91	-47.44	-45.80	-42.53
Nb. Observations	1596	1596	1513	1039	1039	990
Time preference						
Post	0.133***	0.121***	0.121***	0.113***	$0.101^{***}$	0.110***
	(0.022)	(0.022)	(0.024)	(0.028)	(0.028)	(0.030)
Relative effect $(\%)$	20.18	18.63	18.63	16.43	14.94	16.43
Nb. Observations	1604	1604	1517	1043	1043	994
Robust standard errors		$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$
Bootstrapped standard errors	$\checkmark$			$\checkmark$		
Individual fixed-effect	$\checkmark$		$\checkmark$	$\checkmark$		$\checkmark$
Random fixed-effect		$\checkmark$			$\checkmark$	
Time-variant controls		$\checkmark$			$\checkmark$	
Sample: Non-anxious people			$\checkmark$			$\checkmark$

#### Table D2: Temporal variation of preferences (Alternative specifications)

Notes: Short-term variation : on balanced panel between baseline and midline.

Medium-term variation: on balanced panel between baseline, midline and endline.

Mean risk aversion (gain domain) in baseline = 74.53% for the short-term sample and 71.90% for the medium-term sample. Mean risk aversion (loss domain) in baseline = 60.12% for the short-term sample and 61.13% for the medium-term sample. Mean impatience in baseline = 64.41% for the short-term sample and 66.94% for the medium-term sample.

Post is a dummy variable indicating the second wave and the third wave.

Controls include having a job the week before the survey, age, level of education, asset score, household size, subjective health status.

Robust standard errors in parentheses. Significance level: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.